

Review



Three-Dimensional Point Cloud Applications, Datasets, and Compression Methodologies for Remote Sensing: A Meta-Survey

Emil Dumic ^{1,*} and Luís A. da Silva Cruz ^{2,3}

- ¹ Department of Electrical Engineering, University North, 104. brigade 3, 42000 Varaždin, Croatia
- ² Department of Electrical and Computer Engineering, University of Coimbra, 3030-290 Coimbra, Portugal; lcruz@deec.uc.pt
- ³ Instituto de Telecomunicações, 3030-290 Coimbra, Portugal
- * Correspondence: edumic@unin.hr

Abstract: This meta-survey provides a comprehensive review of 3D point cloud (PC) applications in remote sensing (RS), essential datasets available for research and development purposes, and state-of-the-art point cloud compression methods. It offers a comprehensive exploration of the diverse applications of point clouds in remote sensing, including specialized tasks within the field, precision agriculture-focused applications, and broader general uses. Furthermore, datasets that are commonly used in remote-sensing-related research and development tasks are surveyed, including urban, outdoor, and indoor environment datasets; vehicle-related datasets; object datasets; agriculture-related datasets; and other more specialized datasets. Due to their importance in practical applications, this article also surveys point cloud compression technologies from widely used tree- and projection-based methods to more recent deep learning (DL)-based technologies. This study synthesizes insights from previous reviews and original research to identify emerging trends, challenges, and opportunities, serving as a valuable resource for advancing the use of point clouds in remote sensing.

Keywords: point cloud; remote sensing; point cloud datasets; point cloud compression

1. Introduction

Three-dimensional point clouds (PCs) have always attracted substantial attention in remote sensing (RS) because of their ability to accurately represent complex 3D structures and surfaces. Typically, point clouds are represented as a set of distinct points in a threedimensional space (so-called geometric information), possibly with one or more attribute components per point, such as color, reflectance, temperature, or other variables of interest. PC data are often obtained using advanced sensing technologies such as light detection and ranging (LiDAR), photogrammetry, radio detection and ranging (RADAR), synthetic aperture RADAR (SAR), sound detection and ranging (SONAR), and other 3D scanning methods. These types of data are very important and widely used in remote sensing applications such as environmental monitoring, urban planning, forestry, and disaster management, as they provide precise spatial information regarding an object's location and dimensions, as well as terrain topography.

The need for solving problems related to the efficient storage, transmission, and processing of such massive datasets has increased, along with the usage of point cloud data in remote sensing. Point clouds, which are sometimes highly detailed, tend to involve



Academic Editor: Firstname Lastname

Received: 18 January 2025 Revised: 19 February 2025 Accepted: Published:

Citation: Dumic, E.; da Silva Cruz, L.A. Three-Dimensional Point Cloud Applications, Datasets, and Compression Methodologies for Remote Sensing: A Meta-Survey. *Sensors* 2025, 1, 0. https://doi.org/

Copyright: © 2025 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/). enormous amounts of data, which makes their handling and use challenging. To address this problem, efficient point cloud data compression has emerged as an important research area, and various methodologies have been proposed to reduce point cloud data size while preserving geometric accuracy and attribute representation fidelity. Newer compression methodologies based on deep learning (DL) are of special interest as they are a fast-evolving alternative to earlier non-DL compression approaches.

This article is organized into three main sections, describing different aspects of PC applications in RS:

- Section 2: A meta-survey of RS-related PC applications;
- Section 3: PC datasets for RS-related tasks;
- Section 4: PC compression methodologies.

Section 2 provides a review of articles that describes the different PC applications in RS explored in this study. The surveyed articles are organized into three subsections: general PC-related, specific RS-related, and agriculture-related applications. Next, Section 3 introduces several PC datasets used in research and algorithmic development, classified into six categories: urban scenes, outdoor- and vehicle-related contexts, indoor scenarios, small-size and medium-size object representation, agriculture-related contexts, and other application-specific datasets. Section 4 covers different PC compression methods, and they are divided into several categories: common tree-based point cloud compression, projection-based point cloud compression; voxelized or octree-based static or dynamic PC geometry compression; point-based point cloud compression; and other point cloud compression methods and point cloud compression applications. Finally, Section 5 presents conclusions and future research.

2. Point Cloud Applications in Remote Sensing

The following paragraphs present a review of survey articles that describe the applications of PCs in RS activities. A total of 59 survey articles published in several journals in the past 10 years were selected for review using the Scopus database, with the keywords "point cloud" and "remote sensing" applied. Some articles that are not surveys but present original research results were included due to their importance and relevance to this meta-survey.

We also present some information about keyword occurrences and the co-occurrence frequency of the surveyed papers. We used VOSviewer [1] to prepare the keyword occurrence graph shown in Figure 1. We used at least five occurrences (common keywords) from all reviews, except for the "compression" keyword (with four occurrences); combined words with similar meaning; removed the word "review"; and graphically represented the relationships between the 27 selected keywords. Figure 1 illustrates the topics frequently analyzed alongside the searched keywords "point cloud" and "remote sensing". This visualization highlights the relationships between various themes and concepts, listing the areas of research and application commonly associated with these keywords.

It can be concluded that the composite keyword "point cloud" belongs to the same (green) cluster as the 12 keywords: classification, compression, computer vision, dataset, deep learning, image segmentation, learning systems, machine learning, satellite imagery, segmentation, semantic segmentation, and semantics. Similarly, the composite keyword "remote sensing", which belongs to the red cluster, is associated with 13 keywords: 3D computer graphics, 3D modeling, airborne laser scanner, antennas, data acquisition, data handling, extraction, forestry, laser applications, laser scanning, LiDAR, mapping, and photogrammetry.

The keyword "dataset", with five occurrences, discussed in Section 3, is linked to multiple terms across both clusters. In the green cluster, it connects to computer vision, deep learning, and machine learning, among others, while in the red cluster, it is associated with 3D modeling, LiDAR, and remote sensing.

The keyword "compression", with four occurrences, discussed in Section 4, is associated with five keywords from the green cluster—classification, deep learning, point cloud, semantics, and semantic segmentation—and two keywords from the red cluster: remote sensing and LiDAR. It can be observed that compression analysis has received relatively limited attention in review articles focusing on PCs in RS. Therefore, PC compression algorithms will also be reviewed in depth in Section 4.



Figure 1. VOSviewer results, using at least 5 overlapping keywords (except compression keyword) from 59 review articles described earlier in the section, showing a total of 27 keywords. Line width represents the normalized strength of the link between two keywords, i.e., the number of joint keywords from analyzed papers. Circle size represents the weight of the specific keyword, i.e., the number of occurrences in analyzed papers. Different colors are used to represent keyword clustering (two clusters in this case). For more information, consult the observations in [1].

The following subsections discuss several applications, categorized into three groups according to the selected review articles, as shown in Figure 2: general PC-related, specific RS-related, and agriculture-related applications.

Additional research discussing other aspects of the capture and use of PC are reported by the authors of [2], who address data acquisition technologies, intelligent processing algorithms, and their applications in RS in scientific and engineering contexts. Newer collections of research also include two editorials, with the authors of [3] focusing on intelligent PC processing, sensing, and understanding and the authors of [4] exploring PC processing with machine learning techniques. A noteworthy source of information is provided by the authors of [5], who introduce advanced theories and methodologies for AI-driven PC processing, with applications to earth observation, 3D vision, autonomous driving, smart cities, and geospatial information systems.



Figure 2. Map of discussed point cloud applications in remote sensing.

2.1. General Point Cloud-Related Applications

This subsection summarizes different review papers focused on the general applications of PC to RS tasks. The studies surveyed are listed in Table 1, which provides information about the year of publication, type of platform used in the work (groundbased, aerial, etc.) and type of application.

Table 1. Summary of recently published review papers describing general PC processing for RS tasks: MS/HS—multispectral/hyperspectral; SS—semantic segmentation; RGB-D—depth generated from stereo/multiview structure-from-motion (SfM) or depth cameras.

Application	Short Description	Scanner Type	Platform Type	Paper (Y)
General analysis and processing of RS data	General and RS PC processing tasks: scene understanding, compression, and completion	RGB-D, LiDAR	Ground, aerial	[6] (2022)
	General analysis of 2D/3D RS data	RGB-D, LiDAR, SAR, MS/HS	Aerial, satellite	[7] (2022)
	General and RS PC processing tasks: acquisition, processing, and engineering applications	RGB-D, LiDAR, MS/HS	Ground, aerial, spaceborne	[2] (2023)
	Compression methods for automotive LiDAR PC	LiDAR	Ground	[8] (2024)
	Compression methods for automotive LiDAR PC with analysis of impact on object detection	LiDAR	Ground	[9] (2024)

A	Chart Description	C	Distance Trees	D
Application	Short Description	Scanner Type	Platform Type	Paper (Y)
RS applications in scene understanding	PC processing, general, and in RS: feature extraction, object detection, and SS	RGB-D, LiDAR	Ground	[10] (2019)
	PC segmentation, general, and in RS	RGB-D, LiDAR, SAR, MS/HS	Ground, aerial	[11] (2020)
	SS of images and PC	RGB-D, LiDAR, SAR, MS/HS	Aerial, spaceborne	[12] (2021)
	PC segmentation, general, and in RS	LiDAR	Aerial	[13] (2021)
	PC segmentation, general, and in RS	LiDAR	Ground, aerial	[14] (2022)
	PC processing, general, and in RS: classification, detection, and segmentation	RGB-D, LiDAR	Ground, aerial	[15] (2024)
3D mesh processing	PC and mesh processing, general, and in RS: classification, detection, and segmentation	RGB-D, LiDAR	Ground, aerial	[16] (2019)
	SS of 3D meshes	RGB-D, LiDAR	Ground, aerial	[17] (2023)
PC registration	Registration of LiDAR data	LiDAR, MS/HS	Ground, aerial	[18] (2018)
	Image/PC matching in computer vision and RS tasks	Not specified	Not specified	[19] (2023)
	PC registration, general, and in RS	RGB-D, LiDAR	Ground, aerial	[20] (2024)
Multispectral RS data	Fusion of RGB-D and LiDAR data in different RS applications	RGB-D, LiDAR, MS/HS	Ground, aerial, spaceborne	[21] (2017)
	Spectral RS measurements	RGB-D, LiDAR, MS/HS	Aerial	[22] (2018)
	Multispectral LiDAR applications in RS	LiDAR, MS/HS	Ground, aerial, spaceborne	[23] (2024)

Table 1. Cont.

2.1.1. General Analysis and Processing of Remote Sensing Data

The article by Camuffo et al. [6] reviews recent DL-based PC processing algorithms for semantic scene understanding (classification, detection, and semantic segmentation (SS)), compression, and PC completion. In contrast to previous, less structured studies, this study suggests a new taxonomical classification of the methods covered based on variables such as the setup for the acquisition, the properties of the PC data that are acquired, data formatting, side information inclusion, and the features of DL architectures. This classification identifies areas for future research and presents performance evaluations using well-established datasets. The compression models discussed in this study, along with several others, are explained in detail in Section 4.

The authors of [7] examine different applications of computer vision and pattern recognition methods relative to RS data for change detection, boundary extraction, land cover mapping, and target detection. They include a variety of imaging modalities such as digital elevation models (DEMs); LiDAR PCs; and multispectral, hyperspectral, and SAR imagery.

In [2], Yang et al. address current research directions and trends in three areas: point cloud big data acquisition (scanner types and scanner platforms); PC processing (such as denoising, completion, registration, segmentation, and surface reconstruction); and different engineering applications (such as geospatial information, smart cities, underground space development, infrastructure construction, automotive industry, and cultural heritage).

The authors of [8] provide an overview of current techniques for compressing point cloud data from vehicular LiDAR sensors. They present a comprehensive classification that categorizes these methods into four main groups: coding-based, format-based, 2D, and 3D compression. The article evaluates these methods based on key performance metrics such as the compression ratio (CR), bits per point (bpp), and point-to-plane error measures such as the MSE and PSNR.

In [9], Martins et al. characterize the impact of LiDAR PC compression on object detection evaluated in the Kitti dataset [24]. The article presents a thorough review of LiDAR PC compression methods (including learning-based methods) and object detection methods. The compression models used were modified JPEG Pleno PC coding [25], G-PCC [26], and L3C2 [27], and they are described in more detail in Section 4. The authors indicate the availability of the datasets prepared for the study, which can be provided upon request.

2.1.2. Remote Sensing Applications of Scene Understanding

State-of-the-art approaches to mobile laser scanner (MLS) data processing are summarized by the authors of [10]. The tasks reviewed include segmentation based on feature extraction, object detection, SS, and feature extraction (i.e., low-level properties such as edge detection). The benchmark datasets that are currently available for SS and object recognition are also listed.

Point cloud semantic segmentation (PCSS), which extends SS in 2D images to 3D images by employing irregularly dispersed points in 3D space rather than regularly distributed pixels in 2D images, is described by the authors of [11]. Point clouds can be produced from stereo or multiview imagery, or they can be directly acquired via distance-measuring devices. Progress in stereovision algorithms and diverse 3D sensors has enabled the easy generation of 3D point clouds.

Yuan et al., the authors of [12], reviewed recent developments in DL and basic deep neural network designs to perform SS in RS data, including novel data types such as PCs and hyperspectral images. When compared to applications in satellite imaging, recent techniques typically perform poorly on unconventional, unstructured PCs and rich spectral images. According to the authors, learning from very small datasets results in a performance gap, showing that the limited availability of labeled non-conventional RS data presents a major challenge to the development and assessment of novel deep learning techniques.

In [13], two distinct aerial LiDAR datasets are used to conduct a thorough evaluation of three popular DL networks for PCSS: PointNet++, SparseCNN, and KPConv. These networks are assessed for generalization, computation time, classification accuracy, and sensitivity to changes in hyper-parameters.

The authors of [14] review advanced DL models for LiDAR PC segmentation in RS. The authors summarize publicly available 3D datasets for deep learning training and testing and report performance values obtained when benchmarking the methods surveyed on widely used datasets. The authors state that the dynamic graph CNN (DGCNN) and ConvPoint outperform other CNN models in remote sensing applications while maintaining lightweight structures.

Three categories of PCSS techniques—projection-based, voxel-based, and direct pointbased techniques—are presented in detail and compared by the authors of [15], who also provide a comprehensive overview of their development. Within the framework of PCSS, each approach has a distinct use case. It is argued that when high-performance computing systems are not available, projection-based techniques are the best option because they prioritize computational efficiency over performance. According to the authors, voxelbased techniques are appropriate for 3D object classification because they capture the entire context, while point-based methods work well for applications such as 3D SS and are excellent at capturing fine features.

2.1.3. Three-Dimensional Mesh Processing

Recent DL architectures for 3D sensed data processing, including segmentation, object detection, and classification, are reviewed by the authors of [16]. Background ideas, conventional techniques, and contemporary methods and representation modalities such as meshes, RGB-D, multi-view, volumetric, ordered, and unordered point clouds are covered. The study also provides a descriptive list of datasets available for each type of representation. The report finishes with a thorough analysis of deep learning's prospects for processing 3D sensed data, emphasizing the areas that would benefit most from more research.

The article by Adam et al. [17] offers a thorough summary of recent advancements in DL algorithms for SS in 3D meshes representing scenes at the urban scale. Several mesh-based learning methods are described, generalizing DL algorithms on mesh surfaces. Along with a discussion of benchmark large-scale mesh datasets and a comparative analysis of the evaluated approaches, evaluation tools for assessing segmentation performance are provided.

2.1.4. Point Cloud Registration

The authors of [18] provide a thorough evaluation of feature-based coarse registration and fine registration techniques for LiDAR data in photogrammetry and RS. The methods considered are based on coarse features and include surface-based, line-based, and pointbased techniques. The fine registration techniques described are iterative approximation techniques (i.e., iterative closest point), normal distribution transforms (NDTs), random sample consensus (RANSAC [28]), and techniques utilizing auxiliary data. As explained in [29], the RANSAC algorithm can be also used in combination with the scale-invariant feature transform (SIFT) to enhance registration efficiency. A comprehensive review of current advancements in RANSAC-based methods can be found in [30]. The absence of uniform assessment procedures and standard data has been noted as a major drawback as it impedes a fair comparison between methods.

A thorough overview of the concepts and techniques for DL-based mismatch reduction is provided by the authors of [19]. The authors provide an overview of several network designs, geometric information extraction methods, and training modes. The authors list current mining techniques, describe their permutation invariant features, and highlight the significance of permutation invariance in these operations. In order to clarify the principles and efficacy of widely used techniques, both intuitive and mathematical analyses are offered.

An extensive description of DL-based PC registration is detailed by the authors of [20]. This review provides insights from four different angles: the attention mechanism, graph convolutional network, multi-layer perceptron, and deep neural network. Also included is a comprehensive analysis of registration performance measures and datasets related to point cloud registration based on deep learning.

2.1.5. Multispectral Remote Sensing Data

Recent fusion techniques for optical images and LiDAR utilized in photogrammetry and RS are reviewed by the authors of [21]. Many techniques, including real orthophotograph creation, pan-sharpening, key target recognition, registration, classification, change detection, 3D reconstruction, and forest inventory, are presented for data fusion in a variety of applications.

The authors of [22] assess the most recent techniques for unmanned aerial vehicles (UAVs) spectral RS, including geometric processing, sensor technologies, measurement protocols, and radiometric calibration. The authors explain the trajectory of reflected energy

as it travels from particles to being represented as pixels in 3D spectral point clouds, surface maps, or 2D maps.

A comprehensive review of recent multispectral LiDAR technologies and their uses is provided by the authors of [23]. The applications covered include topographic mapping, change detection, ecology and forestry, bathymetry, objects and land use/land cover (LULC) categorization, geology and archaeology, and navigation.

2.2. Remote Sensing in Specific Point Cloud Applications

This subsection provides an overview of several review papers dedicated to specific RS-related PC applications. The articles surveyed are listed in Table 2, which adopts the same format as Table 1.

Table 2. Summary of recently published review papers describing specific RS-related PC applications: MS/HS—multispectral/hyperspectral; BIM—building information modeling; SS—semantic segmentation; RGB-D—depth is generated from stereo/multiview structure-from-motion (SfM) or depth cameras; VR/AR—virtual reality/augmented reality.

Application	Short Description	Main Scanner Type	Platform Type	Paper (Y)
PC to urban model reconstruction and BIM	3D urban model reconstruction from PC	RGB-D, LiDAR	Ground, aerial	[31] (2018)
	3D urban model reconstruction from PC	RGB-D, LiDAR, SAR, MS/HS	Ground, aerial	[32] (2021)
	Building information modeling	RGB-D, LiDAR	Ground, aerial, spaceborne	[33] (2022)
Road detection and extraction in RS	Road information inventory	RGB, LiDAR, MS/HS	Ground, aerial	[34] (2016)
	Road extraction in RS	RGB-D, LiDAR, SAR, MS/HS	Ground, aerial, spaceborne	[35] (2022)
Power line modeling	Power line modeling	RGB-D, LiDAR	Ground, aerial	[36] (2023)
Urban object change detection	Urban object change detection	RGB-D, LiDAR	Ground, aerial	[37] (2023)
Structural damage mapping	Structural damage mapping	RGB-D, LiDAR	Aerial	[38] (2019)
	Critical infrastructure monitoring using LiDAR	LiDAR, MS/HS	Ground, aerial	[39] (2023)
Shoreline mapping	Shoreline mapping	-	Aerial, spaceborne	[40] (2022)
	Shoreline mapping	LiDAR	Aerial	[41] (2023)
Landslide detection	Landslide detection	RGB-D, LiDAR, SAR, MS/HS	Ground, aerial, spaceborne	[42] (2020)
PC segmentation of discontinuous plane surfaces	PC segmentation of discontinuous plane surfaces	RGB-D, LiDAR	Ground, aerial	[43] (2022)
PC SS for specific RS-related tasks	Urban land cover SS	RGB, LiDAR, MS/HS	Aerial, spaceborne	[44] (2015)
	PC SS in heritage building information modelling	RGB-D, LiDAR	Ground, aerial	[45] (2023)
RS and other data analysis in space	RS and other data analysis in space and Mars InSight lander	RGB-D	Space	[46] (2019)
RS and other data analysis in water	RS systems frequently used in ocean research	RGB, LiDAR, SAR, SONAR, MS/HS	Ground, aerial, spaceborne, shipborne	[47] (2022)
	RS systems frequently used in ocean research	RGB, LiDAR, SAR, SONAR, MS/HS	Ground, aerial, spaceborne, shipborne	[48] (2023)
	Airborne oceanic LiDAR RS	LiDAR	Aerial	[49] (2023)
	Autonomous terrain-aided navigation of deep-sea underwater vehicles	RGB-D, LiDAR, Sonar	Underwater	[50] (2024)

Application	Short Description	Main Scanner Type	Platform Type	Paper (Y)
VR/AR applications in RS	Telepresence robots	RGB-D, LiDAR	Ground	[51] (2022)
	"BeHere": collaboration system based on virtual replicas	RGB-D	Ground	[52] (2023)

Table 2. Cont.

2.2.1. Point Clouds in Urban Model Reconstruction and Building Information Modeling

The authors of [31] explore methods that may be used to reconstruct 3D models of urban objects from PCs, including vegetation; buildings; utilities such as electricity lines, roads, and bridges; and free-form architectural features such as statues and curved buildings.

The authors of [32] offer a thorough analysis of cutting-edge point-cloud-based urban scene reconstruction approaches, with a focus on data collection and the advantages and disadvantages of important processing techniques. The authors review various techniques for acquiring, organizing (points, voxels, and patches), registering, and reconstructing point clouds in three dimensions.

The authors of [33] describe the approach of scanning for building information modeling (BIM). Photogrammetry and LiDAR procedures for creating point clouds are covered in the paper. It also compares LiDAR systems mounted on diverse platforms, including airborne, spaceborne, mobile, and terrestrial ones, and discusses the advantages of combining data from several sources. Additionally, thorough explanations of several PC processing techniques—such as registration, sampling, SS, and compression—are given. In addition, compression methods such as SPR-PCC [53] and projection-based algorithms [54] are surveyed, as described later in Section 4.

2.2.2. Road Detection and Extraction in Remote Sensing

An overview of mobile LiDAR technology is provided by the authors of [34], who cover geometrical accuracy validation, data error analysis, direct georeferencing, and system components. A review of studies on road information inventory is carried out with an emphasis on finding and extracting road surfaces, minor structures, and pole-like objects.

The authors of [35] provide a comprehensive review of road extraction techniques using 2D images and 3D LiDAR point clouds. The authors classify these methods into three main categories—2D, 3D, and fused approaches—with additional sub-grouping within each category.

2.2.3. Power Line Modeling

The authors of [36] discuss the benefits and drawbacks of using cutting-edge LiDAR scanning equipment and examine the advantages and disadvantages of several techniques for 3D electrical power line corridor inspection. Their study focuses on techniques for extracting and reconstructing power lines by surveying research articles devoted to that problem, especially those published in conferences and journals related to geosciences. This survey shows that image and PC-based methods are becoming more popular for detecting, locating, segmenting, and inspecting power lines, enabling the automation of tasks related to routine power line inspection and maintenance operations.

2.2.4. Urban Object Change Detection

The most recent advancements in PC data-based urban object change detection are reviewed by the authors of [37]. Thanks to developments in structure-from-motion (SfM) photogrammetry and LiDAR technologies, 3D change detection utilizing PC data has

attracted substantial attention recently. The article offers a thorough examination of applications related to four categories of urban objects: construction sites, street scenes, structures, and urban trees. The evaluation pays more attention to open-source datasets that incorporate change labels and provides an overview of how various data sources are used for each type of object.

2.2.5. Infrastructure Management and Structural Damage Mapping

The evolution of UAV-based structural damage mapping is reviewed by the authors of [38], which moves from basic descriptive overviews to complex texturing and segmentation algorithms and deep learning techniques. Machine learning, enhanced autonomy in mapping, damage mapping in GPS-denied environments, infrastructure maintenance, and robotic UAVs are some of the emerging innovations summarized in this article.

A comprehensive review of LiDAR technology for both commercial and research applications is given in survey [39], along with information on its uses in critical infrastructure monitoring. It covers the monitoring of distribution pipelines for water, oil, and gas; energy production facilities; and ground and air transportation. It also provides an overview of the LiDAR datasets that are currently available for these applications. PCL compression [55], described in Section 4, was proposed as an solution for the problem of PC transmission over Wi-Fi.

2.2.6. Shoreline Mapping

A comprehensive assessment of the literature on shoreline mapping published from 2000 to 2021 is presented by the authors of [40], who attempt to find and analyze research topics and patterns pertaining to shoreline change detection. The authors come to the conclusion that, in light of the significance of safeguarding communities in delta, coastal, and riverine regions, it is imperative to address research gaps in shoreline change analysis by posing new questions and utilizing newer instruments and technologies such as artificial intelligence and machine learning. According to the authors, UAVs, PC data for shoreline change analysis, and high-resolution satellite imagery might all be used as techniques for achieving centimeter-level accuracy.

An overview of coastline mapping using aerial LiDAR is presented by the authors of [41], which covers the availability of data, laser scanning equipment, and current extraction approaches throughout the past 20 years. The authors conclude that there are still data availability issues and some limitations inherent in the technology when it comes to using aerial LiDAR for coastline mapping. Still, many opportunities exist for improvement, particularly when paired with LiDAR point cloud processing methods based on deep learning algorithms.

2.2.7. Landslide Detection

The authors of [42] review typical remote-sensing techniques for landslide assessment, with an emphasis on their applicability to hazard detection and monitoring while taking location and survey costs into consideration. The overview discusses systems that are terrestrial, airborne, and spaceborne and outlines the advantages and disadvantages of each for the acquisition, analysis, and interpretation of data. The presented examples include lasergrammetry, terrestrial optical photogrammetry, and interferometric synthetic aperture RADAR (InSAR).

2.2.8. Point Cloud Segmentation of Discontinuous Plane Surfaces

The benefits, drawbacks, and capabilities of different segmentation algorithms for surface extraction from discontinuity planes are reviewed by the authors of [43], who also discuss the difficulties specific to the processing of PC data representing rock faces.

Analyses of segmentation and orientation results from studies on two rock mass surface PC datasets are presented, and some recommendations for generating consistent and repeatable ground truth orientations are given.

2.2.9. Point Cloud Semantic Segmentation for Specific Remote-Sensing-Related Tasks

The use of small-footprint LiDAR sensors for high-resolution aerial RS applied to urban land cover SS is reviewed by the authors of [44]. The conclusion is that satellite RS has been shown to be effective for monitoring land cover on a wide scale; however, more research has to be carried out on finer-scale maps, particularly in metropolitan areas, as this has been demonstrated by a number of biophysical and socioeconomic studies. The article also covers the use of compression methods in the applications surveyed, namely, LASzip [56], LCMP [57], lossy LAS [58], and octree-based compression [59] (some of which are reviewed in Section 4).

The integration of PCSS with the workflow of historical building information modeling (HBIM) is reviewed by the authors of [45], in a survey study. The article summarizes a few dozen studies covering automatic and semi-automatic methods and tools for geometric modeling applied to HBIM.

2.2.10. Space Exploration and Remote Sensing Applications

The authors of [46] examine the sensor choice and application of the Mars InSight Lander (Interior exploration utilizing Seismic Investigations, Geodesy, and Heat Transport). Image products are used extensively in many lander tasks, such as processing raw telemetry; making mosaics; creating terrain meshes; stereo correlation; radiometric correction; and producing various products such as instrument deployment maps, surface normals, PC data, and layers for roughness maps.

2.2.11. Remote Sensing in Aquatic Environments

The authors of [47,48] cover twelve distinct RS systems that are frequently used in ocean research: four passive (optical systems, thermal infrared radiometers, microwave radiometers, and global navigation satellite system reflectometry) and eight active (SAR, scatterometers, altimeters, LiDAR, gravimeters, SONAR, high-frequency RADAR, and marine RADAR) systems. A thorough evaluation and discussion were conducted on 15 applications of RS in the ocean, utilizing various RS systems and approaches: ocean surface wind, ocean wave height, ocean surface current, ocean tide, ocean surface salinity, ocean color, ocean chlorophyll, ocean oil spills, underwater ocean, sea level, sea ice, icebergs, sea surface temperature, ship detection, and fisheries.

The authors of [49] provide an overview of airborne oceanic LiDAR RS technology and applications. Multi-channel airborne LiDAR devices are intended to greatly enhance the resolution and quality of data for marine biological and geographic profiles. In order to encourage further study in ocean biogeochemistry, algorithms for biological product retrieval and modeling based on common radiation transfer models are described.

The study presented in [50] marks a significant advancement in the autonomous and precise operation of deep-sea autonomous underwater vehicles (AUVs) near the seabed, focusing on enhancing underwater terrain-aided navigation (TAN) techniques. TAN leverages underwater terrain features as reference points for positioning. It enables the real-time localization of AUVs within pre-existing terrain maps by actively detecting and tracking distinct terrain characteristics, maintaining positioning errors constrained within temporal and spatial domains. The article explores the background, operational principles, and key technical aspects of underwater TAN. It reviews the algorithms central to the two primary modules of TAN: the terrain-aided positioning module and the iterative filtering estimation module.

2.2.12. Virtual and Augmented Reality for Remote Sensing Applications

An application of virtual reality/augmented reality (VR/AR) applications in RS involves telepresence robots, which are increasingly recognized for their role in enhancing social interactions [51]. An example of remote collaborative systems called "BeHere", where co-presence can be useful for instructions based on virtual replicas, combining gestures and avatars for procedural tasks, is explained by the authors of [52]. In this case, RGB-D frames are encoded and transmitted to the remote side, where they are decoded and reconstructed into PCs.

2.3. Agriculture-Related Applications

This subsection surveys and synthesizes the contents of several review studies focused on the use of PCs and related representations in remote sensing applications to agriculture. The studies are listed in Table 3 in the usual format.

Table 3. Summary of recently published review papers describing agriculture-related PC processing in RS: MS/HS—multispectral/hyperspectral; RGB-D—depth generated from stereo/multiview structure-from-motion (SfM) or depth cameras.

Application	Short Description	Main Scanner Type	Platform Type	Paper (Y)
General agriculture	Agriculture	RGB-D, LiDAR, MS/HS	Aerial	[60] (2019)
	Agriculture	LiDAR	Ground, aerial, satellite	[61] (2023)
Forestry	Delineation of individual tree crowns	RGB-D, LiDAR, MS/HS	Aerial	[62] (2017)
	Forest management in Nordic countries	RGB-D, LiDAR, MS/HS	Aerial, satellite	[63] (2018)
	Forest RS using drone and LiDAR	LiDAR	Aerial	[64] (2019)
	SfM photogrammetry for RS data in forestry	RGB-D	Aerial	[65] (2019)
	RS analysis of European aspen in boreal forests	RGB-D, LiDAR, MS/HS	Ground, aerial, satellite	[66] (2020)
	Tree species' classification	LiDAR, MS/HS	Aerial	[67] (2021)
	Timber assortments	LiDAR	Ground, aerial	[68] (2022)
	LiDAR versus destructive harvesting to quantify above-ground biomass	LiDAR	Ground	[69] (2022)
	Forest restoration in RS	RGB-D, LiDAR, MS/HS	Ground, aerial, satellite	[70] (2022)
	Forest resource assessment	RGB-D, LiDAR, MS/HS	Ground, aerial, satellite	[71] (2024)
Vegetation parameter extraction	Soil moisture and plant water stress	Thermal image	Satellite	[72] (2016)
	Leaf area index (LAI) retrieval	LiDAR	Aerial	[73] (2021)
	Extraction of vegetation parameters in savanna biome	LiDAR, MS/HS	Ground	[74] (2021)
	Vegetation cover fraction (fCover)	RGB-D, LiDAR, MS/HS	Ground, aerial, satellite	[75] (2023)
Viticulture	Precision viticulture	RGB-D, LiDAR, MS/HS	Ground, aerial, satellite	[76] (2023)
Weed management	Weed management in RS	RGB-D, LiDAR, MS/HS	Ground, aerial	[77] (2022)

2.3.1. General Use

The authors of [60] discuss the use of unmanned aircraft systems (UASs) to affect remote imaging for improved farming operations such as field mapping, chemical spraying, biomass estimation, plant stress detection, weed control, and inventory counting. Different tools and technologies, such as PCs, vegetation indices, machine learning algorithms, and statistical methods, are crucial to precision agriculture.

The authors of [61] examine research on the use of LiDAR systems, such as MLSs, aerial laser scanners (ALSs) and terrestrial laser scanners (TLSs), in precision agriculture, with a focus on crop cultivation. Subsequently, they showcase current LiDAR uses, particularly in digitizing trees and plants, estimating crop-related metrics, planning and decision assistance, and object detection and navigation.

2.3.2. Forestry

Several articles are specific to forestry applications of remote-sensing-based on PCs.

The methods for delineating individual tree crowns from 3D data and their applications in ecology and forestry are reviewed by the authors of [62]. It is determined that while approaches utilizing the entire point cloud are required to identify smaller trees beneath the canopy, 2D surface model methods (derived from point clouds) are often the best for detecting taller trees.

The current status of national forest inventory and forest management in the Nordic countries is examined by the authors of [63], who also highlight the advantages and disadvantages of different RS materials and data-gathering techniques from the viewpoints of various audiences.

The authors of [64] evaluate the technical prerequisites for generating high-quality measurements from autonomous platforms with various drone aircraft and commercial laser scanners. An example of an autonomous helicopter in a Southern Czech Republic's temperate mountain forest is also included in the case study.

The possible application of consumer-grade cameras and unmanned aerial vehicles for terrestrial SfM-based surveys in forestry is covered by the authors of [65]. The authors show that with the help of the SfM workflow, foresters can gather several RS datasets, using a single sensor to generate multiple spatial products.

The authors of [66] summarize current knowledge about the ecological significance of the European aspen, talk about the challenges associated with understanding the species's occurrence and dynamics in boreal forests, and look at the possibilities presented by different RS technologies for aspen mapping.

The authors of [67] summarize research on the categorization of tree species using data from aerial laser scanning, identifying the best classification algorithms and the most useful features generated from LiDAR. It is argued that the most accurate features are those derived from full-waveform data, while radiometric features mixed with height data also work well. Furthermore, according to the publications reviewed, the best results for species discrimination are obtained with support vector machines and random forest classifiers.

Alvites et al., in [68], look at the classification and quantification of timber assortments using terrestrial and aerial LiDAR devices, including UASs. When it comes to describing understory trees, terrestrial LiDAR systems perform quite well. For comprehensive timber assortment data over huge forest areas, combining terrestrial technologies with airborne/UAS LiDAR looks promising. Furthermore, there is increased interest in these approaches, as observed via the increasing usage of machine and DL algorithms in analyzing LiDAR data.

The authors of [69], Demol et al., compare data from ten TLS-derived above-ground biomass (AGB) investigations with values based on destructive tree harvesting. It was concluded that AGB obtained from TLS closely matches values from destructive analyses.

The authors of [70] examine feasible alternatives for developing targets for landscape forest restoration that take spatial patterns into account. The hierarchical levels of a forested surrounding are represented via spatial patterns in analyses. The division of the landscape is carried out hierarchically: sub-catchments, vegetation patches, and individual trees.

The authors of [71] describe distinct instances of DL methods in diverse forest applications and classify them based on their processing techniques and operational principles. Several sensors and equipment used to collect data on forests are introduced. The authors also list and provide details about forest imagery datasets that are currently accessible and investigate the global geographic distribution of the related research.

2.3.3. Vegetation Parameter Extraction

Using satellite thermal images, the authors of [72] investigate the merits and limitations of the most widely used models for determining plant water stress and soil moisture. They also report a number of indicators, such as the normalized differential vegetation index, also applied to assess soil moisture, in addition to evapotranspiration.

Basic leaf area index (LAI) retrieval techniques, validation procedures, and constraints are reviewed by the authors of [73], employing point cloud data from aerial LiDAR scanners. The gap fraction model and empirical regression are the two primary LAI retrieval method types evaluated. The empirical and gap fraction models' poor scalability over time, space, and various airborne LiDAR systems is demonstrated via empirical validation.

The authors of [74] investigate different facets of vegetation parameter extraction with TLS, such as retrieval techniques and parameters extracted from TLS point clouds. Primary and secondary vegetation parameters are examined. The primary parameters are computed directly from point clouds, whilst the secondary parameters are approximated from the primary ones.

The common applications of vegetation cover fraction (fCover) in a variety of fields, settings, and scales are presented by the authors of [75]. Along with traditional non-imaging techniques, the review includes LiDAR return-based techniques (e.g., return intensity retrieval, return number index), image-based techniques (e.g., spectrum retrieval, unmixing, segmentation), and PC-based techniques (e.g., rasterization) on different platforms.

2.3.4. Viticulture

Precision viticulture potential and uses are covered by the authors of [76]. The explanation of various sensor types and their working principles covers both proximal and RS platforms, such as satellites and UAVs. The study includes descriptions of vegetation status indicators used in viticulture, as well as supervised and unsupervised techniques for image classification and segmentation. Additionally, it investigates photogrammetric techniques for dense PC-based 3D canopy modeling. The study also highlights how largescale datasets may be processed and analyzed using deep learning and machine learning approaches to evaluate the physiological and agronomic biomarkers of vineyards.

2.3.5. Weed Management

Dobbs et al. examine the applications of 3D imaging technologies in [77], such as photogrammetry, SfM, and LiDAR, in weed management. The authors explore the applications of 3D photogrammetric imaging in weed management, such as weed detection and mapping for targeted removal, and in weed-related research, such as modeling weed–crop competition, to predict yield loss. They also examine the use of 3D imaging for weed management in orchards and grasslands.

15 of 45

3. Point Cloud Datasets for Remote-Sensing-Related Tasks

This section provides an overview of various datasets that are particularly relevant to tasks related to RS. These datasets are diverse in nature and are categorized based on the type of PC data. To facilitate a more structured understanding, we classify the datasets into six distinct categories, as depicted in Figure 3:

- Urban scenes: This category includes datasets of urban environments, such as buildings, roads, and public infrastructure. These datasets are used for applications such as urban planning or traffic management.
- Outdoor- and vehicle-related contexts: Datasets in this category cover large-scale outdoor environments and vehicle-centric data, often used in autonomous driving research and outdoor navigation systems. They typically include data from roads, highways, and surrounding landscapes.
- Indoor scenarios: These datasets focus on enclosed environments such as homes and offices. They are typically used for applications in robotics and interior mapping.
- Small-size and medium-size object representation: This category is dedicated to datasets capturing smaller objects and simple shapes. It is used for object recognition, segmentation, and shape understanding.
- Agriculture-related contexts: Datasets here represent agricultural landscapes, including crops and fields. These are typically used in precision agriculture and crop monitoring.
- Other application-specific datasets: This category includes datasets that do not fit into the above categories. These datasets are commonly used in research studies on compression methods and visual quality evaluations.

Each of these datasets acts as a reference for the compression methods that will be discussed in further detail in Section 4. Categorization also helps determine which dataset is most suited for particular RS-related tasks, improving the effectiveness of compression methods designed for each type of PC data.

Table 4 summarizes 10 urban scenario datasets, referencing studies from 2017 to 2024, including one online repository, that are frequently referenced in research for evaluating and comparing different algorithms. Each dataset is accompanied by a brief description to highlight its key features and applications. It can be observed that most datasets are primarily designed for semantic segmentation. The UseGeo dataset [78] can be used for 3D reconstruction from images, supporting both single-image depth estimation and multi-view 3D reconstruction, with LiDAR data serving as a reference. Other tasks can be also performed, including image orientation, feature extraction and matching, the automated registration of images with LiDAR data, and semantic segmentation. The 3DTK repository [79] offers a diverse collection of point clouds, primarily intended for testing and developing PC registration algorithms. However, some datasets within the repository can also be utilized for change detection and object detection. Several PCs include additional attributes such as color, thermal, and reflectance information.

Table 5 summarizes 21 outdoor- and vehicle-related datasets, referencing papers from 2011 to 2024. A short description of each dataset is also given. Global navigation satellite system (GNSS) and inertial measurement unit (IMU) information is present in most datasets, except for the Waymo and ONCE datasets, which provide similar information: vehicle poses at different points in time (both) and translational and angular velocity (Waymo).



Figure 3. Map of discussed point cloud datasets for remote-sensing-related tasks.

Table 4. Summary of recently published studies describing urban-level datasets: N. CL—number of classes; N. P—number of points; RSM—real/synthetic/mixed; TLS—terrestrial laser scanner; MLS—mobile laser scanner; ALS—aerial laser scanner; UAV-P—UAV photogrammetry; UAV-L—UAV LiDAR; SS—semantic segmentation; IS—instance segmentation; DE—depth estimation.

Dataset Name	Platform	N. CL	N. P	Area (m ²) or Length (m) Covered	Average Point Density (pts/m ²)	RSM	Task Suitability	Short Description
3DTK [79]	Several	-	_	_	_	R	PC registration	Repository for 3D point clouds from robotic experiments
Semantic3D [80]	TLS	8	$4 imes 10^9$	-	Varying	R	SS	Wide range of urban outdoor scenes
Toronto-3D [81]	MLS	8	$78.3 imes 10^6$	$1\times 10^3 \text{ m}$	1000	R	SS	PC dataset of Toronto, Canada
DALES [82]	ALS	8	$505 imes 10^6$	$10\times 10^6\ m^2$	50	R	SS	Dayton-annotated LiDAR earth scan (DALES)
DALES objects [83]	ALS	8	492×10^6	$10\times 10^6 \; m^2$	50	R	SS, IS	DALES dataset [82] with additional intensity and IS
Swiss3DCities [84]	UAV-P	5.2	$\begin{array}{c} 3147 \times 10^6 \\ \text{hi-res;} \\ 226 \times 10^6 \\ \text{mid-res} \end{array}$	$2.7 \times 10^{6} \text{ m}^2$	1166 hi-res; 84 mid-res	R	SS	PCs from three cities in Switzerland: Zurich, Zug, and Davos
SensatUrban [85]	UAV-P	13 (31)	2847×10^6	$7.64\times10^6\ m^2$	373	R	SS	PCs from three cities in UK: Birmingham with SS, Cambridge with SS, and York without SS
BuildingNet [86]	mesh, PC	31	200×10^6	_	-	S	SS, classification	Labeled mesh and PC building parts; 2000 objects $(100 \times 10^3$ points per mesh) with 292×10^3 annotated components

Dataset Name	Platform	N. CL	N. P	Area (m ²) or Length (m) Covered	Average Point Density (pts/m ²)	RSM	Task Suitability	Short Description
STPLS3D [87]	UAV-P	9/9/9/9 real; 6/17/20 synthetic	-	$\begin{array}{l} 1.27\times 10^6 \text{ m}^2\\ \text{real;}\\ 16\times 10^6 \text{ m}^2\\ \text{synthetic} \end{array}$	100 real; 11 synthetic	М	SS, IS	SS on four real datasets (nine different classes each); synthetic 3D data generation method with three IS datasets (6/17/20 different classes each)
UseGeo [78]	UAV-L, UAV-P	_	$392.8 imes 10^{6}$	$5 0.715 \times 10^{6} \text{ m}^{2}$ (image area)	51	R	Multiview DE, monocular DE, feature extraction and matching, SS	UAV-based multi-sensor (RGB, LiDAR) datasets for geospatial research

Table 4. Cont.

Table 5. Summary of recently published papers describing outdoor- and vehicle-related datasets:RSM—real/synthetic/mixed; bbox—bounding boxes; Det—detection; Tra—tracking; MF—motionforecasting; SS—semantic segmentation; PS—panoptic segmentation; IS—instance segmentation.

Dataset Name	Туре	RSM	Short Description
Ford [88]	LiDAR, 360° RGB	R	3 sequences with 1500 scans each; on average, 100×10^3 points per scan.
KITTI [24]	LiDAR, RGB	R	Det 2D/3D: 7481/7518 train/test images/PC, 80,256 bbox for 3 (3) classes; Tra: 917 tracked objects for 2 classes.
Argoverse [89]	LiDAR, 360° RGB, stereo	R	Det 3D: 22×10^3 scenes, 993×10^3 bounding boxes, 15 (17) classes; Tra: 113 scenes, each 15–30 s, 11,052 tracks, 15 (17) classes; MF: 324,557 scenes, each 5 s, 11.7×10^6 unique tracks, 1 class; Stereo: 6624 stereo pairs with ground truth depth.
nuScenes [90]	LiDAR, 360° RGB RADAR	R	Det 2D/IS 2D: 93,000 images, 800 \times 10 ³ bbox and IS masks (foreground objects), 23 classes; Det 3D: 40 \times 10 ³ PC, 1.4 \times 10 ⁶ bbox, 10 (23) classes.
SemanticKITTI [91]	LiDAR	R	Det 3D: $23 \times 10^3/20 \times 10^3$ train/test PC; 682×10^3 bbox for 8 classes; SS 3D: $23 \times 10^3/20 \times 10^3$ train/test PC; $4,549 \times 10^6$ points for 25 (28) classes.
Waymo [92]	LiDAR, 360° RGB	R	Det 2D/Tra: 1×10^6 images, 9.9×10^6 bbox, 256×10^3 unique IDs, for 3 classes; Det 3D/Tra: 230×10^3 PC, 12×10^6 bbox, 113×10^3 unique IDs, for 4 classes; Added: MF (103,354 scenes, each 20 s, 10.8×10^6 , 3 classes), 2D video (100×10^3 images) SS for 28 and PS for 3 classes, 3D SS for 23 classes.
Argoverse2 [93]	LiDAR, 360° RGB, stereo	R	Det 3D/Tra: 1000 scenes, each 15s with 30 classes; "LiDAR" (unannotated): 20,000 scenes, each 30 s, with LiDAR, HD maps, pose; MF: 250,000 scenes, each 11 s, 10 classes, 13.9×10^6 unique tracks; Map change: 1000 scenes, each 45 s, with LiDAR, HD maps; 200 with map changes.
ONCE [94]	LiDAR, 360° RGB	R	Det 2D: images from annotated PC, 769×10^3 bbox, 5 classes, unannotated 7×10^6 images; Det 3D: annotated 16,000 PC, 417×10^3 bbox, 5 classes, unannotated 1×10^6 PC.
PandaSet [95]	LiDAR, 360° RGB	R	Det 3D: 8240 annotated PC from 103 scenes, 28 classes; SS 3D: 6080 annotated PC from 76 scenes, 37 classes.
Panoptic nuScenes [96]	LiDAR	R	SS 3D/PS 3D/Panoptic tracking: 40×10^3 PC, 1, 1 × 10 ⁹ points for 16 (32) classes.
DAIR-V2X [97]	LiDAR, RGB	R	Sensory inputs from vehicles, infrastructure, and collaborative vehicles–infrastructure: Det 2D/3D: 71,254 images/PC from vehicles and infrastructure, 10 classes, 1.2×10^6 bbox; SS 2D/3D: 71,254 images/PC from vehicles and infrastructure, 10 classes.
ONCE- 3DLanes [98]	LiDAR, 360° RGB	R	Annotated lanes in 2D and 3D from ONCE dataset [94].
Ithaca365 [99]	LiDAR, RGB	R	Repeatedly recorded: diverse scenes, weather, time, and traffic conditions: 2D: bbox, amodal IS, and road segmentation, 7000 images, 6 classes; Det 3D: 175 PC, 6 classes
V2X-Sim [100]	LiDAR, 360° RGB-D	S	Simulated multi-agent perception dataset for collaborative autonomous driving, up to 5 vehicles and 1 set of infrastructure; SS 2D: 6 RGB cameras with 60×10^3 images, bird's eye view; Det 3D/Tra: 10×10^3 PC with 26.6×10^3 bbox.
OpenLane- V1 [101]	LiDAR, 360° RGB	R	Annotated lanes in 2D and 3D from Waymo dataset [92].
OpenLane- V2 [102]	LiDAR, 360° RGB	R	Annotated lanes in 2D and 3D from Argoverse2 [93] and nuScenes [90] datasets.

	Tab	ole 5. Ca	ont.
Dataset Name	Туре	RSM	Short Description
V2X-Seq [103]	LiDAR, RGB	R	Det 3D/Tra: 15,000 images/PC from vehicle and infrastructure, 10 classes, 110 tracked objects per scene, 95 scenes; MF: 210,000 scenes, each 10s, 8 classes, with traffic light.
V2V4Real [104]	LiDAR, 360° RGB	R	Real multi-agent perception dataset for collaborative autonomous driving, 2 vehicles with collaboration; Det 3D: 40×10^3 images, 20×10^3 PC with 240×10^3 bbox, 5 classes.
Occ3D [105]	LiDAR	R	3D voxel occupancy dataset semi-automatically labeled from Waymo [92] and nuScenes [90] datasets.
V2X-Real [106]	LiDAR, 360° RGB	R	Real multi-agent perception dataset for collaborative autonomous driving, 2 vehicles and 2 sets of infrastructure with 4 collaboration combinations; Det $3D$: 171×10^3 images, 33×10^3 PC with 1.2×10^6 bbox, 10 classes.
WaterScenes [107]	4D RADAR, RGB	R	2D: bounding box, pixel annotations, 7 classes, 54,120 objects; 3D: point-level RADAR PC annotations, 7 classes, 54,120 objects; Tasks: object detection, waterline segmentation, free-space segmentation, object tracking, SS, IS, PS, panoptic perception.

Table 6 summarizes 13 indoor-scene datasets, referencing papers from 2015 to 2023, that are frequently utilized in research articles for benchmarking and evaluating various algorithms. Each dataset is accompanied by a concise description, highlighting its primary characteristics and typical applications.

Table 6. Summary of recently published papers describing indoor-scene datasets: N. CL—number of classes; N. SC—number of scenes; RSM—real/synthetic/mixed; PS—panoptic segmentation; IS—instance segmentation.

Dataset Name	Туре	N. CL	N. SC	RSM	Short Description
SUN-RGBD [108]	RGB-D,	800	10,355	R	Semantic annotation of 10,355 RGB-D scene images in 47 scenes
	PC				categories with about 800 object categories; annotated 146,617 2D polygons and 64,595 3D bounding boxes
S3DIS [109]	PC	12	5	R	Stanford 3D indoor scene; semantic annotation of five indoor-area PCs with 215 \times 10 6 points
SceneNN [110]	mesh, RGB-D	40	100	R	100 RGB-D video scenes, reconstructed, annotated with per-vertex and per-pixel labels, bounding boxes for 1482 objects, object poses
Scannetv2 [111]	mesh, RGB-D	20	1,613	R	1,513 RGB-D video scenes with 3D camera poses, reconstructed, 36,213 objects (18 classes) with bounding boxes; voxel segmentation (18/20 classes for IS/PS, respectively)
SUNCG [112]	mesh	84	49,884	S	49,884 valid floors, with contain 404,058 rooms and 5,697,217 object instances from 2644 unique object meshes covering 84 categories; used for the semantic scene completion
Structured3D [113]	mesh, RGB	_	3500	S	3D "primitive + relationship" structure annotations of 21,835 rooms in 3500 scenes; 196,000 photo-realistic 2D renderings of the rooms
ARKitScenes [114]	PC, RGB-D	_	5047	R	RGB-D and PC acquisition of 5047 indoor scans; PC with annotated object bounding boxes from 17 furniture categories
HM3D [115]	mesh; RCB	-	1000	R	Habitat–Matterport 3D: 1000 building-scale textured 3D mesh
Scannet200 [116]	PC	200	1513	R	3D IS based on ScanNet with 200 classes
TO-Scene [117]	mesh	52	16,077	М	16,077 scenes with real tables and 60,174 synthetic objects on; vertex segmentation, 3D bounding boxes, and camera poses
Scannet++ [118]	PC, RGB-D	>1000	460	R	460 3D reconstructions of indoor scenes with dense semantic and instance annotations; DSLR images and RGB-D sequences
HM3DSem [119]	mesh; RGB	1625	216	R	Habitat–Matterport 3D Semantic: 142,646 object instance annotations of 216 3D spaces with 3100 rooms
Instruct3D [120]	PC	_	280	R	280 scenes from Scannet++ [118] with approximately 10 different segmentation instructions, with 2565 instruction–point cloud pairings

Table 7 summarizes eight datasets featuring 3D objects, referencing papers from 2015 to 2023, that are widely used in research articles to benchmark and compare different

algorithms. Each dataset is accompanied by a brief description, summarizing its key features and typical research applications.

Table 7. Summary of recently published papers describing object datasets: N. CL—number of classes;N. O.—number of objects; RSM—real/synthetic/mixed; IS—instance segmentation.

Dataset Name	Туре	N. CL	N. O.	RSM	Short Description
ModelNet [121]	mesh	660	151,128	S	Annotated per model class
ShapeNetCore [122]	mesh	55	51,300	S	Annotated per model class
ShapeNetSem [122]	mesh	270	12,000	S	Annotated per model class; additional information is present
ShapeNet Parts [123]	mesh	16	31,963	S	Annotated parts, 42 labels for 16 classes, from ShapeNetCore dataset
Visionair [124]	mesh	_	60	S	Sixty distinct models, from rigid items (i.e., Chair) to smooth non-rigid objects (i.e., Bunny), downloaded from the Visionair repository
ScanObjectNN [125]	PC	15	15,000	R	From 700 scenes from SceneNN and Scannet selected 2902 unique objects; IS of 15 categories (with part annotations)
PartNet [126]	mesh	24	26,671	S	573,585 part instances with fine-grained, instance-level, and hierarchical 3D part information
OmniObject3D [127]	mesh; video	190	6000	R	Input meshes are rendered to PC and RGB-D images are included; COLMAP camera poses; rich text description of each object
Objaverse-XL [128]	mesh	_	>10 × 10 ⁶	М	10.2 million 3D deduplicated objects, coming from several sources, including metadata information, i.e., textual description

Table 8 summarizes six datasets related to agriculture that are frequently referenced in research articles, referencing papers from 2020 to 2024. Each dataset is briefly described, highlighting its core features and specific agricultural applications and providing a clear understanding of their suitability for various research tasks.

Table 8. Summary of recently published studies describing agriculture-related datasets (N. CL number of classes; N. P/F—number of points/faces; ULS—UAV laser scanner; MLS—mobile laser scanner; SS—semantic segmentation).

Dataset Name	Platform	Туре	N. CL	N. P/F	Short Description
LFuji-air [129]	MLS	PC	-	-	11 LiDAR-based PCs of Fuji apples trees with 1353 apple annotations
Corn50 [130]	_	PC	_	_	50 RGB PCs of artificial corn plants
H3D [131]	ULS	PC, mesh, RGB	11	73,909,354 P 8,550,338 F	LiDAR PCs and meshes of the village of Hessigheim, captured at four different epochs
Pheno4D [132]	ULS	PC	3	260×10^6	PCs from 7 maize and 7 tomato plants over several days; Segmentation of "soil", "stem", and instance "leaf" points
VineLiDAR [133]	ULS	PC	_	356,633,530 P	10 3D LiDAR PCs in LASzip [56] format with RGB color
Treescope [134]	ULS, MLS	PC	_	-	SS and diameter estimation in agricultural environments: pine, oak, maple, and cedar forests; almond and pistachio orchards

Table 9 summarizes 12 specific datasets, referencing studies from 2016 to 2023, including one online repository, that are frequently utilized in research studies focused on compression techniques and visual quality assessments. Each dataset is briefly described, showcasing its relevance and use cases in the context of PC data processing and evaluation.

Dataset Name	Туре	N. O	RSM	Short Description
JPEG Pleno [135]	PC, mesh	-	R	Diverse set of static and dynamic PCs for different tasks, such as static and dynamic PC compression
MVUB [136]	PC	10	R	JPEG Pleno Database: Microsoft Voxelized Upper Bodies (MVUB)— A Voxelized Point Cloud Dataset (dynamic)
8iVFB v2 [137]	PC	4	R	JPEG Pleno Database: 8i Voxelized Full Bodies (8iVFB v2)— A Dynamic Voxelized Point Cloud Dataset with 10-bit depth
Owlii [138]	PC	4	R	Owlii Dynamic Human Textured Mesh Sequence Dataset, 4 dynamic PCs
8iVSLF [139]	PC	6	R	8i Voxelized Surface Light Field (8iVSLF) Dataset— A Dynamic Voxelized Point Cloud Dataset with 12-bit depth
ICIP2020 [140]	PC	6	R	Static 6 original from [135] and 90 processed PCs, using two compression algorithms (G-PCC, V-PCC [141]) and octree pruning, for objective quality assessment
vsenseVVDB2 [142]	PC, mesh	8	R	Dynamic 4 PCs and 4 mesh sequences, compressed with Draco (for meshes) [143], G-PCC and V-PCC (for PCs) codecs [141] 152 distorted; for objective quality assessment
SJTU-PCQA [144]	PC	10	R	Static 10 original from [135] and 420 processed PCs, using 7 distortion types, for objective quality assessment
UVG-VPC [145]	PC	12	R	Dynamic voxelized PCs for visual volumetric video-based coding
BASICS [146]	PC	75	R	Static 75 original and more than 1200 processed PCs using 4 compression algorithms, for objective quality assessment
PC-plosses [147]	PC	4	R	Dynamic 3 original (from [135,138]) and 105 processed PCs, V-PCC-compressed [141] and degraded by packet losses, for objective quality assessment
WPC [148]	PC	20	R	Static 20 original and 740 processed PCs, using 5 distortion types, for objective quality assessment

Table 9. Summary of recently published papers describing specific datasets (N. O—number of objects;RSM—real/synthetic/mixed).

4. Point Cloud Compression

This section provides an overview of some of the current PC compression models and methods, with the aim of providing additional information to fill the gaps left by most of the review papers covered in this survey. Two articles surveyed in this text, refs. [6,8], are exceptions in this aspect as they describe and compare some point cloud compression methods. An important aspect of PC compression methods is the evaluation of the distortion introduced by the (lossy) coding, usually through the use of objective measures that compare the quality of the reconstructed PC with that of the original PC. A good amount of information about these measures can be found in [149,150].

We classify the compression methods into several categories with some common properties, as shown in Figure 4, in which the main categorization is related to the basic coding principle. While general lossless data compression algorithms such as ZIP or RAR can be applied to PCs, they are not specifically designed for this purpose and are therefore not included in this survey.

Table 10 provides an overview of the PC compression methods and their specific application types and use cases inferred mostly from the datasets employed in the reviewed studies. The methods listed are compared relative to several properties: the type of point cloud (static and dynamic), point cloud components (geometry and attributes), and compression fidelity (lossless and lossy).

The "Dynamics" column in Table 10 indicates whether the method is applied to static, dynamic, or both types of PCs, based on the tested cases in the reviewed studies. Methods labeled with only "static" could potentially be used in dynamic scenarios but do not exploit temporal redundancy between PCs. Conversely, only "dynamic" signifies that the model is explicitly designed for dynamic PCs, considering temporal redundancies. The "L/LS" (lossy/lossless) column in Table 10 indicates whether the method can be used for lossy or lossless compression fidelity or both. In some cases, both methods can be used.



Figure 4. Map of discussed point cloud compression algorithms (PCC-point cloud compression).

Table 10. Summary of PC compression methods and their applications: PCC—point cloud compression; S—static; D—dynamic; G—geometry; A—attribute; L/LS—lossy/lossless; VR/AR—virtual reality/augmented reality.

PCC Group	Model	Dynamics	PC Type	L/LS	PC Application (best for)	
Common	G-PCC [26]	S,D	G,A	L,LS	Autonomous driving, cultural heritage	
tree-based	ased GeS TM [151]		G,A	L,LS	VR/AR (solid PCs)	
	Octree-based [59]	S	G,A	LS	Urban scenario	
	PCL [55]	S	G,A	L,LS	Not specific	
	CWI-PCL [152]	S,D	G,A	L	VR/AR, telepresence, "MPEG-anchor" codec	
	AVS-PCC [153]	S,D	G,A	L,LS	Autonomous driving, cultural heritage, VR/AR	
	Draco [143]	S	G,A	L,LS	VR/AR, telepresence, PC and mech compression	
Projection-based	Projection-based [154]	S	G,A	L	Urban scenario	
	Projection-based [54]	S	G	L,LS	Autonomous driving	
	Projection-based [155]	D	G,A	L	VR/AR	
	V-PCC [156]	S,D	G,A	L,LS	VR/AR, telepresence	
	L3C2 [27]	S	G,A	L,LS	Autonomous driving	
Voxelized	DSAE [157]	S	G	L	VR/AR	
static	ADAE [158]	S	G	L	VR/AR, buildings	
geometry	PCGCv1 [159]	S	G	L	VR/AR	
	PCGCv2 [160]	S	G	L	VR/AR	
	SparsePCGC [161]	S	G	L,LS	Autonomous driving, VR/AR	
	PCGformer [162]	S	G	L	VR/AR	
	Geo-CNN v1 [163]	S	G	L	VR/AR	
	Geo-CNN v2 [164]	S	G	L	VR/AR	
Octree-based	OctSqueeze [165]	S	G	L	Autonomous driving	
static	VoxelDNN [166]	S	G	LS	VR/AR	
geometry	MSVoxelDNN [167]	S	G	LS	VR/AR	
	SibContext [168]	S	G	L	Autonomous driving	
	OctFormer [169]	S	G	L	Autonomous driving, indoor scenes	
	Octree-Retention [170]	S	G	L	Autonomous driving, VR/AR	
Voxelized	Dynamic PCGCv2 [171]	D	G	L	VR/AR	
dynamic	D-DPCC [172]	D	G	L	VR/AR	
geometry	Dynamic PCC [173]	D	G	L	VR/AR	
	Dynamic SparsePCGC [174]	D	G	L,LS	VR/AR	
Octree-based	MuSCLE [175]	D	G	L	Autonomous driving	
dynamic	VoxelContext-Net [176]	S,D	G	L	Autonomous driving, indoor scenes	
geometry	OctAttention [177]	S,D	G	L,LS	Autonomous driving, VR/AR	
	STAEM [178]	S,D	G	L,LS	Autonomous driving, VR/AR	
	EHEM [179]	D	G	L,LS	Autonomous driving	
Point-based	RNN-based [180]	S	G	L	Autonomous driving, urban scenario	
	AE-based [181]	S	G	L	Simple objects	
	CACTUS [182]	S	G	L	Autonomous driving	
	3QNet [183]	S	G	L	Autonomous driving, indoor scenes, objects	
	IPDAE [184]	S	G	L	Objects, indoor scenes, autonomous driving	
	SPR-PCC [53]	S	G	L	Autonomous driving	
	Pointsoup [185]	S	G	L	Indoor scenes, autonomous driving (trained only on objects)	
Voxelized geometry	JPEG Pleno PCC [25]	S	G,A	L	VR/AR, buildings, cultural heritage, urban scenario	
and attribute	CNeT [186]	S	G,A	LS	VR/AR	
	Unicorn [187,188]	S,D	G,A	L,LS	VR/AR, autonomous driving, indoor scenes, buildings	

PCC Group	Model	Dynamics	PC Type	L/LS	PC Application (best for)
Attribute	Deep-PCAC [189]	S	А	L	VR/AR, cultural heritage, buildings, indoor scenes
	MNeT [190]	S	А	L,LS	VR/AR
	ScalablePCAC [191]	S	А	L	VR/AR
	ANF-based [192]	S	А	L	VR/AR (trained only on indoor scenes)
	Model [193]	S	А	LS	Objects, indoor scenes, VR/AR, autonomous driving
	Model [194]	D	А	L	VR/AR
Neural	NVFPCC [195]	S,D	G	L	VR/AR
radiance	LVAC [196]	S	А	L	VR/AR
field	Model [197]	S	G,A	L	VR/AR, urban scenario
	Model [198]	D	G,A	L	VR/AR
	ResNeRF-PCAC [199]	S	А	L	VR/AR
Other PCC	LASzip [56]	S	G,A	LS	LiDAR LAS PC compression
methods	MrSID [200]	S	G,A	L,LS	LiDAR LAS PC compression

Table 10. Cont.

The next subsections describe each of the mentioned compression methods in more detail, while the last subsection discusses PC compression applications.

4.1. Common Tree-Based Point Cloud Compression

To construct the compressed point cloud, G-PCC [26] encodes the content directly in 3D space. Geometry and attribute data are encoded independently in G-PCC. Since geometry is necessary for attribute coding, geometry coding is firstly applied. Coordinate transformation, voxelization, and either an octree or a trisoup ("triangle soup") surface approximation are the first steps in the geometry encoding process. Finally, to attain lower bitrates, arithmetic coding is used. Three choices are offered for attribute coding: a lifting transform, a predicting transform, and the region-adaptive hierarchical transform (RAHT). Upon applying one of these transforms, the coefficients undergo quantization and arithmetical encoding. The authors of [141,201,202] contain further information regarding G-PCC.

Other prior studies based on octree can be also found, i.e., [59], for lossless compression. The point cloud library (PCL) [55] offers an octree-based method for compressing point clouds.

The CWI-PCL ("MPEG anchor") codec is discussed in [152], based on the point cloud library (PCL) [55], for dynamic PC geometry and attribute compression, and it is suitable for 3D real-time tele-immersion. Newer methods based on G-PCC have also been recently proposed, such as G-PCC++ [203], which addresses compression distortion and improves quality compared to the original G-PCC. In another paper based on G-PCC [204], the authors propose a solution to optimize the octree codec of G-PCC to be more precise. GeS TM is also proposed [151] as a branch derived from the G-PCC test model for compressing static and dynamic solid PCs.

Another open-source library called Draco [143], based on kd-tree, can also be used to compress and decompress PCs, as well as 3D geometric meshes.

The authors of [205] propose an approach that investigates redundancies between the successive frames of a dynamic PC sequence in order to reduce geometry information in a lossless manner. It operates by gradually increasing the octree's resolution. Experimental results show better performance than Draco and CWI-PCL (in a lossless mode), for example, using MVUB [136] and 8iVFB v2 [137] datasets.

AVS (Audio Video Coding Standard Workgroup of China) established a subgroup in 2019, AVS-PCC, to efficiently compress LiDAR point clouds that are both static and dynamic

in different scenarios: autonomous driving, cultural heritage, and dynamic scenarios [153]. Recently, geometry and attribute bit rate allocations were optimized for precise bit control by the authors of [206]. More details about AVS-PCC are provided by the authors of [201].

4.2. Projection-Based Point Cloud Compression

Using V-PCC coding [156], the PC is first divided into multiple connected regions to generate 3D surface segments. Such 3D surface segments are called "patches" and are subsequently projected one at a time into a 2D patch. Reducing projection issues such as occlusions and hidden surfaces is made easier using this technique. Each 2D patch consists of a set of images: a binary image, a geometry image (depth map), and attributes of the projected points. H.265/HEVC video compression is employed to compress the 2D sequences with the packed patches after they are generated, although any other compression method may also be utilized. More details about V-PCC are provided by the authors of [141,201].

Projection-based techniques were applied previously, as demonstrated by the authors of [154], who presented a projection-based PC compression algorithm. The authors of [54] introduce a projection-based PC compression strategy comprising four techniques: range image conversion, PC segmentation, prediction, and coding. These methods were evaluated using the KITTI dataset [24]. The authors of [155] developed a new dynamic PC compression method that combines surface reconstruction with various projection types and bit depths with latter video compression to produce geometry and texture maps. Recently, some papers also proposed improvements for V-PCC. The authors of [207] propose voxel selection-based refining segmentation to accelerate the PC-refining segmentation process, as well as data-adaptive patch packing to reduce occupancy map size. The authors of [208] propose an efficient geometry surface coding method to improve geometry information compression in V-PCC. The authors of [209] propose a lightweight, fully connected network-based fast CU size decision during H.265/HEVC utilization for V-PCC. The authors in [210] propose a method for segmenting dynamic point clouds based on shape similarity and occlusion before generating patches. The experimental results indicate that the proposed method outperforms V-PCC and some other existing methods for both geometric and texture data.

A low-latency, low-complexity codec (L3C2) was recently proposed by MPEG [27]. L3C2 was developed for the rotating LiDAR sensor, storing points such as a coarse and then residual 2D projection in polar coordinate systems. It can be used for both lossy and lossless coding, as well as for geometry and attribute compression.

4.3. Voxelized Static Geometry Point Cloud Compression

The authors of [157] present geometry-based compression called DSAE (Distributed Source AutoEncoder), which divides input data into $8 \times 8 \times 8$ voxel blocks. These blocks are then represented by the encoder using a deep syndrome (which corresponds to the autoencoder's hidden variables). The decoder then reconstructs the coded PC using coded features (deep syndrome data) and side information from the side information encoder. An improved version of PC geometry compression is presented by the authors of [158] using adversarial distributed source autoencoders.

Another study [161] comprises an improved version of PCGCv1 and PCGCv2 [159, 160], called SparsePCGC. In SparsePCGC, the convolutions are only carried out on sparsely distributed, most probably positively occupied voxels. The proposed model uses a sparse convolution-based neural network (SparseCNN) and occupancy probability approximation model based on SparseCNN in order to calculate the occupancy probability in a single-stage or in a multi-stage manner. SparsePCGC can be used for both lossless and lossy geometry compression. The proposed approach demonstrates excellent performance across a variety

of datasets, including sparse LiDAR PC geometry (SemanticKITTI [91] for training and testing and the Ford dataset [88] for testing) and dense object PC geometry (JPEG Pleno dataset [135,137], Owlii dataset [138], and MVUB dataset [136]; Shapenet dataset [122] for training), compared to G-PCC and other DL-based models while having low complexity. The same authors propose lossy PC geometry compression in [162] using transformer networks stacked with sparse convolutions, showing better results than G-PCC.

The authors of [163] provide a lossy PC geometry compression technique called Geo-CNN for static point clouds that uses uniform quantization and learned convolutional transforms. An improved version, Geo-CNN v2, is also presented [164].

4.4. Octree-Based Static Geometry Point Cloud Compression

The authors of [165] propose OctSqueeze, which initially encodes LiDAR points into an octree, an effective data structure appropriate for PCs with sparse points. The probabilities of the octree symbols are then modeled in a conditional entropy model with a tree structure, which encodes the octree into a compact bitstream. The experimental results are compared with Draco [143] and CWI-PCL codec [152] using two datasets, the newly created NorthAmerica (proposed by the same authors) and SemanticKITTI [91] datasets, showing better results for the proposed codec.

The authors of [166] demonstrate a context-adaptive arithmetic coding-based DL lossless compression technique for static PC geometry, called VoxelDNN. The proposed encoder functions in a hybrid mode that combines voxel-based and octree coding. The experimental results on ModelNet [121], MVUB [136], and 8iVFB v2 [137] show better results compared to G-PCC v12. A multiscale approach, MSVoxelDNN, is presented in [167], speeding up encoding and decoding times compared to VoxelDNN.

An innovative multi-level framework based on octrees is presented in [168] for large-scale sparse and unstructured PC compression. By utilizing the context of neighbors, ancestors, and siblings' children, the framework employs a novel entropy model to explore hierarchical dependency in an octree. Experimental results using the SemanticKITTI [91] and nuScenes [90] datasets show better results compared to G-PCC, Draco, and VoxelContext-Net.

The authors of [169] propose OctFormer, an octree-based transformer compression technique that does not rely on sibling nodes' occupancy data. The proposed approach builds octree node sequences using non-overlapping context windows and distributes the outcome of a multihead self-attention operation over multiple nodes. Experiments using the SemanticKITTI [91] and ScanNet [111] datasets show the better performance of the Oct-Former model compared to G-PCC and OctSqueeze. VoxelContext-Net (without coordinate refinement models, i.e., postprocessing) exhibits a similar performance to OctFormer on the ScanNet dataset and somewhat lower performances with respect to SemanticKITTI for higher bitrates, and performance enhancements that are several times higher are also observed. OctAttention exhibits similar reconstruction quality on the SemanticKITTI dataset; however, OctFormer's decoding time is much faster.

The authors of [170] present the Octree-Retention model. Initially, the point cloud objects are segmented using an octree structure. Then, important features are extracted from sibling and ancestor nodes using octree-based contextual windows. Finally, prior knowledge between spatially nearby nodes can be successfully used for compression using the Octree-Retention model, which uses retentive networks (RetNet). Experiments show better performance compared to a) G-PCC, VoxelContext-Net, OctAttention, and OctFormer for the SemanticKITTI [91] dataset (lossy mode) and b) G-PCC, VoxelDNN, and OctAttention for the 8iVFB v2 [137] dataset (lossless mode).

4.5. Voxelized Dynamic Geometry Point Cloud Compression

The authors of [171] present dynamic PC geometry compression using variational autoencoders with temporal autoregressive hyperprior and sparse convolutions, with the PCGCv2 model [160] used for individual PCs. Improved geometrical quality is realized compared to the V-PCC described earlier. Using 3D motion estimation, inter-frame geometry PC coding, and compensation in the feature space, the authors of [172] suggest a unique 3D sparse convolution-based deep dynamic point cloud compression (D-DPCC) network to compress and adjust the dynamic PC geometry, also showing better performance than V-PCC inter-frame coding. This is tested on the 8iVFB v2 human body dataset [137]. The authors of [174] use a multiscale sparse representation (MSR) framework from static PCs to compress dynamic PC geometry, advancing the static SparsePCGC encoder [161]. The suggested approach achieves lower bpp values in comparison to G-PCC and SparseGCPC in a lossless mode, and it realizes BD-rate gains in the lossy mode compared to V-PCC, SparsePCGC, PCGCv2, D-DPCC, and the methods reported by the authors of [173].

4.6. Octree-Based Dynamic Geometry Point Cloud Compression

To save storage space for LiDAR sensor data streams, the authors of [175] describe a compression algorithm, MuSCLE, which takes advantage of spatiotemporal relationships across many LiDAR sweeps. A new conditional entropy model is suggested to represent the likelihood of the octree symbols by considering coarse-level geometry and geometric and intensity information from previous sweeps. Afterwards, the complete data stream is encoded into a compact one. The experimental results using the SemanticKITTI [91] and NorthAmerica (proposed by the same authors) datasets show better performances compared to the OctSqueeze, Draco, CWI-PCL, and G-PCC codecs.

The authors of [176] suggest VoxelContext-Net, a two-stage deep learning system, for both dynamic and static point cloud compression. The suggested method combines the advantages of voxel-based schemes and octree-based techniques by compressing octree structured data using the voxel's context. The experimental results using two datasets, SemanticKITTI [91] (static and dynamic case) and ScanNet [111] (static case), compared with OctSqueeze, G-PCC, and Draco, show better results for the proposed codec.

The authors of [177] present OctAttention, a multi-context deep learning codec that makes use of the memory-efficient octree structure for point clouds. Through the collection of sibling and ancestor nodes, the proposed method encodes octree symbol sequences. Experiments that use static LiDAR PC via SemanticKITTI [91] show the better performance of the proposed model in the lossy mode compared to VoxelContext-Net (without coordinate refinement models, i.e., postprocessing) and Octsqueeze. Moreover, using dynamic MVUB [136] and 8iVFB v2 [137] results in better outcomes in the lossless dynamic case compared to VoxelDNN, MSVoxelDNN, G-PCC v1, and the model reported by the authors of [205].

The authors of [178] suggest a learning-based entropy model, STAEM (Spatiotemporal Attention Entropy Model), for dynamic PC compression that takes advantage of the large-scale spatiotemporal context based on octrees. The authors provide a graph-based feature extraction methodology that takes geometry into account in order to extract useful features from a large-scale, informative context. Moreover, a spatiotemporal attention mechanism is presented by the authors to identify dependencies in the large-scale context. The experimental results with respect to SemanticKITTI [91] (static and dynamic cases) show better results, compared to OctAttention, OctSqueeze, and G-PCC in the static case and G-PCC and MuSCLE in the dynamic case. Moreover, using dynamic MVUB [136] and 8iVFB v2 [137] results in better outcomes in the lossless dynamic case compared to VoxelDNN, G-PCC, and OctAttention.

To improve the efficiency of the octree-based auto-regressive entropy model, the authors of [179] suggest an EHEM (efficient hierarchical entropy model), a hierarchical attention structure that preserves the global receptive field and exhibits linear complexity relative to the context scale. In addition, the authors provide a grouped context structure that maintains compression efficiencies while resolving the auto-regression-related serial decoding problem. The experimental results using the SemanticKITTI [91] and Ford [88] datasets show better performances in the dynamic case compared to the OctAttention, SparsePCGC, and G-PCC models while exhibiting a decoding latency that is similar to the effective conventional G-PCC model.

The authors of [211] provide spherical coordinate-based learning PC compression (SCP), a model-agnostic technique that takes advantage of the point clouds' many azimuthal angle invariance features and circular shapes. Furthermore, in order to reduce the reconstruction error for far-off regions inside the spherically coordinate-based octree, the authors suggest a multi-level octree for SCP. The experimental results that use the SemanticKITTI [91] and Ford [88] datasets show better performances when using SCP with EHEM and OctAttention compared to the baselines, SparsePCGC, and G-PCC.

4.7. Attribute Point Cloud Compression

Several studies have dealt with PC static attribute compression. The authors of [189] present end-to-end deep lossy point-based PC attribute compression, called Deep-PCAC, assuming that the geometry is coded with some existing geometry codecs. A multiscale lossless (or lossy) voxelized PC attribute coding method called MNeT is presented in [190], assuming lossless geometry coding.

The authors of [212] present an approach called SparsePCAC that uses sparse convolutions to create a variational autoencoder (VAE) framework for compressing PC attributes, assuming lossless geometry coding. The experimental results show that SparsePCAC performs better than G-PCC v6 and existing DL methods. The authors of [191] propose a scalable PC attribute compression method called ScalablePCAC, assuming lossless geometry coding. It uses G-PCC as the base layer and a model as an enhancement layer, showing better performances than G-PCC v14, v22, and SparsePCAC.

The authors of [192] provide a PC attribute compression strategy based on the augmented normalizing flow (ANF) model, including sparse convolutions and assuming lossless geometry coding. In comparison to VAE-based coding schemes, the normalizing flow model's invertibility allows for improved reconstruction. The experimental results show better performances compared to G-PCC v14, Deep-PCAC, and SparsePCAC.

The authors of [213] present an embedded attribute PC encoding method based on SPIHT as an alternative to the RAHT transform within G-PCC. Furthermore, the authors of [214] propose a scalable, embedded PC attribute encoding based on a multilayer perceptron used with an RAHT transform within G-PCC.

The authors of [193] present lossless point-based PC attribute compression, assuming lossless geometry compression, and this was tested using a diverse set of PCs: objects (i.e., ShapeNet [122]), indoor scenes (i.e., ScanNet [111]), JPEGs (i.e., 8iVFB v2 [137]), and LiDAR (SemanticKITTI dataset [91]). Better performance was observed compared to G-PCC, MNeT [190], and CNeT [186].

An end-to-end learned dynamic lossy attribute coding point-based method is presented in [194], assuming lossless geometry compression, which uses effective highdimensional convolution to capture complex inter-point correlations. The experimental results show better results compared to region-adaptive hierarchical transform (RAHT) attribute compression models within the G-PCC codec.

4.8. Voxelized Geometry and Attribute Point Cloud Compression

The JPEG Pleno Point Cloud Coding codec was recently introduced in [25], and it is currently used as Verification Model V4.0 [215,216], which has joint geometry and an attributecoding system. Additionally, the optional DL module performs upsampling/superresolution to enhance the final quality of decompressed PCs. Details about the training and test PCs for the comparison between submitted models can be found in [149]. The training dataset consists of the ShapeNet dataset [122] and different PCs, both static and dynamic, from JPEG or MPEG providers, while in the test dataset, there are 12 defined PCs with three types of PCs present: solid, dense, and sparse. These represent different density classes. Lossless static PC geometry and attribute compression models (CNeT and Context NeTwork) were presented in [186] and tested on human-body datasets (i.e., 8iVFB v2 [137], Owlii [138]).

Unicorn [187,188] is a newer learning-based solution designed to compress static and dynamic, geometrical, and attribute PCs with different source characteristics (such as 8iVFB v2 [137], Owlii [138], JPEG Pleno [135], KITTI [24], Ford [88], and Scannet [111] datasets) in both lossy and lossless modes. It realizes significantly better performance than standard methods such as MPEG G-PCC, V-PCC, and other learning-based approaches, delivering state-of-the-art compression efficiency with a practical level of complexity.

4.9. Point-Based Point Cloud Compression

Several studies have been conducted that use direct PCs as inputs. A method utilizing a recurrent neural network and residual blocks to gradually compress the data from a single frame of 3D LiDAR is presented by the authors of [180].

Other papers have proposed autoencoders for generative purposes; however, they may also be used for compression-related tasks. For example, the authors of [181] use autoencoder models to learn the compact representation of a PC, while generative adversarial networks (GANs) (with both raw PC data and latent space data) and Gaussian mixture models (GMMs) (with latent space data) are studied to generate novel PCs.

The authors of [182] propose a coding system called Content-Aware Compression and Transport Using Semantics (CACTUS), which divides the original PC into independent streams using SS with the RandLA-Net [217] architecture. The segments are then encoded with DSAE, G-PC, or Draco compression.

A novel DL-based PC compression model is proposed in [183], called the 3D PC Geometry Quantization Compression Network (3QNet), which can handle dense points; it can overcome the existing point-based approaches' robustness issue. Experiments related to Visionair [124], ScanNet [111], and SemanticKITTI [91] show that 3QNet is capable of achieving better compression efficiencies than CWI-PCL, G-PCC, Draco, PCGCv2, and Geo-CNN.

The authors of [184] propose an IPDAE model (improved patch-based deep autoencoder), which incorporates several improvements over the patch-based point cloud compression method described by the authors of [218] (inspired from PointNet [219]). These consist of octree coding for centroid point sampling, a learnable context model for entropy coding, and an integrated training and compression procedure. The experimental results using ModelNet [121], ShapeNet [122], and S3DIS [109] show better outcomes compared to G-PCC, PCGCv2, Geo-CNN, and Geo-CNN v2.

The authors of [53] propose scene-aware LiDAR PC geometry compression using semantically prior representation (SPR-PCC) by projecting PCs to 2D images, segmenting projection images, and eventually removing moving objects from a set of projected frames (depending on the final application task). The proposed solution shows better results compared to G-PCC v14 and PCL using the KITTI dataset [24].

The authors of [185] present Pointsoup, an effective learning-based geometry codec that simultaneously realizes both very low decoding latencies and high performances. A point model-based approach that uses dilated window-based entropy modeling and an attention-based encoder inspired by the traditional Trisoup codec was introduced. The experimental results using S3DIS [109], ScanNet [111], and SemanticKITTI [91]—compared to G-PCC v23 (octree and trisoup), OctAttention, IPDAE, and 3QNet—showed better results for the proposed codec.

4.10. Neural Radiance Field Point Cloud Compression

The novel-view synthesis approach NeRF (neural radiance field) [220], which is also applicable to PC compression, is covered in this subsection. NeRF was first proposed to reconstruct a 3D scene from sparse 2D representations, i.e., to synthesize novel views of a scene. NeRF approaches can be classified into three types based on how they depict the scenes: implicit, explicit, and hybrid [221]. Explicit and hybrid radiance field representations alleviate the slowness of implicit representations by including explicit data structures (such as 2D/3D grids or 3D points) for local feature encoding. The same study introduces binary radiance fields (BiRFs), a storage-efficient hybrid model of radiance fields that uses binary feature encoding. Another storage-efficient NeRF method was recently proposed by the authors of [222], called context-based NeRF compression (CNC), following the design of state-of-the-art binary radiance field (BiRF) compression. The experimental results on the Synthetic-NeRF (synthetic) [220] and Tanks and Temples (real) [223] datasets showed the better performance of the proposed CNC method compared to the BiRF.

PC geometry compression using NeRF was presented by the authors of [195], called NVFPCC, for both static and dynamic PCs. The authors of [196] present learned volumetric attribute compression (LVAC) for PCs using coordinate-based networks. The authors of [197] present a unified approach for geometry and attribute static PC compression using NeRF. Two networks were used for the geometry and attribute components. The voxelized body dataset from the authors of [137] were used to carry out comparisons with existing solutions, namely, the G-PCC standard and NVFPCC, showing better performances for the proposed solution. PCs from the Semantic3D dataset [80] were also compared to those reported by the authors of [160,162], showing better performances when using geometry quality assessments. The authors of [198] propose an end-to-end pipeline for volumetric video compression utilizing neural-based representation. Three-dimensional dynamic content is represented as a sequence of NeRFs, which are converted from explicit to neural representations. The experimental results show better performances than G-PCC and Draco, and similar performances are observed relative to the V-PCC coding solutions using the 8iVFB v2 [137] and 8iVSLF [139] datasets. The authors of [199] present a new method for point cloud attribute compression called residual neural radiance fields for point cloud attribute coding (ResNeRF-PCAC). Tests on the 8iVFB v2 [137] dataset show better results than G-PCC; the region-adaptive hierarchical transform (RAHT); and pred/lift schemes for attribute coding, v14 and v21.

4.11. Other Point Cloud Compression Methods

In this subsection, we explore several additional compression methods that are not addressed in the previously discussed subsections.

The lossless compression scheme for LiDAR in the binary LAS (LASer) data format, LAZ (LASzip), is presented by the authors of [56,224], achieving 7-20% of the uncompressed size. The LASzip compressor is lossless, non-progressive, streaming, and order-preserving, and it provides random access. Another lossless LiDAR PC compression method, the LAS compression coder (LCMP), was presented by the authors of [57,225], with only 10–20% of

their original size. Lossy LAS file compression using uniform space division was presented by the authors of [58]. A commercially available compression format multi-resolution seamless image database (MrSID) is also available for images and LAS LiDAR files for both lossless and lossy compression [200].

4.12. Point Cloud Compression Applications

In this subsection, we discuss different PC compression applications, mostly based on the employed datasets, as described in Table 10. The most common PC applications comprise virtual reality/augmented reality (VR/AR) with MVUB [136], 8iVFB v2 [137], and Owlii [138] datasets, as well as autonomous driving with the KITTI [24] and SemanticKITTI [91] datasets. For VR/AR applications, trained datasets are usually sampled objects from the ModelNet [121] and ShapeNet [122] datasets, which means that those models should also work for these datasets. In the case of the ANF-based model for attribute compression [192] and Pointsoup for geometry compression [185], the dataset used in model training is mentioned because its type is completely different from the test datasets; this is carried out to test their generalization abilities in compressing different PC types.

Less prevalent tested datasets include indoor scene datasets (usually ScanNet [111]) and urban scenario datasets (i.e., Semantic3D [80], 3DTK [79]), used only in a few proposed compression algorithms. Regarding the datasets with object PCs, the sampled ModelNet [121] and ShapeNet [122] datasets are usually used to train DL-based models.

In non-DL based models, standardized codecs such as G-PCC [26], V-PCC [156], and AVS-PCC [153] are versatile, supporting both static and dynamic PCs with lossless and lossy options and several application types. Specialized compression tools such as LASzip [56] and MrSID [200] focus on LiDAR PC compression. Draco [143] can be used in telepresence and VR/AR due to efficient geometry and attribute encoding. The newly proposed L3C2 codec can be used in autonomous driving.

From Table 10, it can also be observed that DL-based models combining geometry and attribute compression are less common compared to geometry-only models. Notable examples include lossless CNeT [186], lossy JPEG Pleno PCC [25], and lossy models [197] for static PCs, as well as lossy models [198] for dynamic PCs.

It is also notable that some DL-based methods either omit the exact G-PCC version used for comparison or rely on older versions available at the time of evaluation. This can be attributed to the rapid development of the G-PCC and V-PCC codecs, with the latest versions possibly not being publicly accessible. Additionally, comparisons often focus only on similar codec types, overlooking other DL-based methods. Since different codecs are tailored to specific PC types, as indicated by the datasets they use (e.g., dense PCs such as 8iVFB v2 [137] or sparse PCs such as SemanticKITTI [91]), it is essential to consider and specify the exact PC type when comparing methods against the G-PCC codec.

A summary of the advantages and disadvantages of each PC compression group described earlier is provided in Table 11.

PCC Group	Advantages	Disadvantages		
Common tree-based PCC	 General category of many older and newer tree-based PCC models G-PCC represents newer standardized model [26] with rapid development 	Depends on specific model		
Projection-based PCC	 General category of many older and newer projection-based PCC models V-PCC represents newer standardized model [156] with rapid development 	Depends on specific model		
Voxelized static geometry PCC	Better suited for dense PCs, i.e., in VR/AR applications	 Works only for geometry PCs Temporal redundancies are not taken into account Input PC needs to be voxelized 		
Octree-based static geometry PCC	• Better suited for large-scale sparse PCs, i.e, in autonomous driving applications	 Works only for geometry PCs Temporal redundancies are not taken into account Decoding complexity 		
Voxelized dynamic geometry PCC	Better suited for dense PCs, i.e., in VR/AR applications	Works only for geometry PCsInput PC needs to be voxelized		
Octree-based dynamic geometry PCC	• Better suited for large-scale sparse PCs, i.e., in autonomous driving applications	Works only for geometry PCsDecoding complexity		
Attribute PCC	In some cases, can be used instead or on top of existing solutions such as G-PCC	• Usually, lossless geometry compression is assumed		
Voxelized geometry and attribute PCC	 Newer solutions usually based on deep learning models, as alternative to stan- dardized codecs such as G-PCC and V- PCC 	• Training data have to be carefully chosen (i.e., depending on bit depth and motion for dynamic PCs)		
Point-based PCC	 Input PC does not have to be voxelized Generally works better for unevenly distributed and sparse point clouds 	Complexity depends on number of points		
NeRF PCC	 Any volumetric input data can be used Images rendered with NeRF do not have visual artifacts due to PCs' discrete nature Can be used to compress plenoptic PCs 	Slower training timeFewer plenoptic PC datasets		
Other PCC methods	• Specific use cases for LiDAR PCC with specific type of PCC algorithm	Slower loading performance using com- pressed LAZ compared to uncompressed LAS, but also depends on software used [226]		

Table 11. Summary of advantages and disadvantages of reviewed PC compression methods (PCC—point cloud compression).

4.13. Point Cloud Compression Limitations and Research Trends

As summarized in Table 11, the PC compression methods reviewed have some drawbacks. In general, the compression efficiency in lossy modes is still not very high, especially when high fidelity (i.e., low geometrical distortion) is needed, as in the case of remote sensing applications involving some metrology operations. At low and medium coding rates (i.e., low bits per point), well-established methods based on geometry processing such as octree-decomposition can introduce point position errors that are incompatible with applications in which accurate measurements are to be carried out from decoded/reconstructed point clouds. To address this problem, research is ongoing with respect to the compression of point clouds in order to attain near-lossless performance where the distortion allowed is measured via the min–max criteria (i.e., ensuring bounded maximum positional errors computed over all point cloud points) instead of the currently used average point-to-point or point-to-surface errors. Deep-learning-based solutions, despite showing promising performance and, in some cases, surpassing that of alternative solutions, also exhibit specific problems, such as the introduction of points representing artificial structures during reconstruction/decompression, possibly impacting the operation of downstream processing. Learning-based solutions also suffer from complexity ails, usually requiring fast and energy-expensive parallel processing hardware for acceptable coding and decoding timecomplexity, somewhat limiting the contexts in which they can be deployed. Concerning

scalability and random access functionalities, which are necessary, for instance, to decompress and render the representations of compressed 3D point clouds at different scales and/or in different parts of the encoded point clouds without decoding the entire point cloud, in general, all methods developed thus far only support these functionalities in a limited fashion. Most methods, both deep-learning-based and non-deep-learning-based, resort to block partitioning and encoding in order to provide some sort of random access, allowing the independent decoding of each block. In the case of scalability, some compression methods such as MPEG G-PCC and V-PCC support some scalability, but most deep-learning-based methods have no provisions for scalable coding. A notable exception is the method described by the authors of [187], which can support scalable encoding and decoding. Thus far, most research on point cloud compression has focused on the efficient representation of geometrical information, which is, without a doubt, the most important component for most remote sensing applications. However, attribute information that can represent surface temperatures, surface reflectances, etc., is also very important and quite hard to represent efficiently. Recognizing this, several research groups (e.g., those reported in [188]) and standard organizations such as ISO/IEC JPEG and MPEG are working on this side of the point cloud compression problem. It is expected that the results of these research efforts will translate into more compact compressed 3D point clouds that are easier to store and process, with advantages for remote sensing applications and processing workflows.

5. Conclusions and Future Research

This article is organized into three main sections: a meta-review of review papers on the RS application of point clouds (Section 2), datasets commonly used in RS-related algorithm research and development (Section 3), and PC compression methods (Section 4).

Section 2 surveys a selection of review articles about the applications of PC technology in RS-related contexts, with articles divided into three groups: general PC-related, specific RS-related, and agriculture-related applications. The first group of articles cover PC acquisition and processing tasks, such as scene understanding, compression, segmentation, registration, multispectral applications, and multimodal data fusion. Some of these processing tasks are not exclusive to remote sensing and are also used in, e.g., computer vision applications relative to several problems. The second and third groups of this article are more specifically related to RS tasks, with the third group further divided for agriculture-related applications. Tables 1–3 summarize each discussed article. The range of applications covered in the section, as well as the number of studies selected for review, shows that point clouds are used in many application scenarios that fall under the umbrella of remote sensing.

Section 3 provides a list of datasets (point clouds and related) used for the research of PC compression methods. Section 4, as well as research in other areas, refers to the following topics: autonomous vehicle navigation (outdoor- and vehicle-related contexts), robotics and interior mapping (indoor scenarios), precision agriculture (agriculture-related contexts), and other application-specific datasets for visual quality evaluations and related objective quality measures. Many of the mentioned datasets are also used in PC object detection, as well as semantic, instance, and panoptic segmentation. Datasets containing 2D RGB (+depth) images are also suitable for image segmentation tasks and for the fusion of 2D and 3D data. More detailed information regarding each dataset can be found in Tables 4–9 and the articles referenced in those tables. Overall, this section shows that there is now a reasonably large set of 3D point cloud and related dataset formats, and in many cases, they are accompanied by application-specific annotations, which are very useful for anyone wanting to carry out research on remote sensing problems.

Section 4 provides information regarding different PC compression methodologies that researchers have developed; these methodologies aim to provide efficient point cloud data transmission, handling, and storage that are usable in remote sensing applications. The methods surveyed are divided into several categories: common tree-based, projection-based, and other methods. The methods are able to process voxelized or unprocessed real coordinate point clouds in both static or dynamic scenarios. Besides PC geometry component compression, some methods also address attribute compression. Emerging representation formats such as neural radiance fields and Gaussian splattings are also covered. Several standardized point cloud coding methodologies, such as MPEG's geometry-based (G-PCC) and video-based (V-PCC) codecs, and recent DL-based models, such as the JPEG Pleno Point Cloud Coding codec, are included in this survey.

As point cloud technology evolves, it is expected that new datasets, algorithms, and hardware improvements will further enhance its application in remote sensing technologies and systems. Future research should concentrate on creating even more effective compression techniques that balance data volume and representation fidelity, in addition to investigating how deep learning models may help optimize point cloud data processing. Compression algorithms for both geometry and attribute data (potentially utilizing radiance-field synthesis) could be explored, considering different application contexts such as urban scenarios, indoor scenes, autonomous driving on water surfaces (i.e., the WaterScenes dataset [107]), and agriculture-related environments. Both lossy and lossless compression methods can also be explored. New algorithms could be tested using both real and synthetic datasets created via generative tools, such as Objaverse-XL [128], with respect to both real and synthetic objects. Due to their flexibility and economy of representation, point clouds will continue to be essential for improving remote sensing capabilities and applications as they are used to tackle the issues identified in this survey.

Author Contributions: Conceptualization, E.D.; methodology, E.D. and L.A.d.S.C.; writing—original draft preparation, E.D.; investigation, E.D. and L.A.d.S.C.; and writing—review and editing, L.A.d.S.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: No new data were created in this manuscript.

Acknowledgments: This work was funded by FCT/MECI by project reference 10.54499/UIDB/50008/2020, and DOI identifier https://doi.org/10.54499/UIDB/50008/2020 and when applicable co-funded by EU funds under UID/50008: Instituto de Telecomunicações.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

The following abbreviations are used in this manuscript:

PC	Point cloud
PCC	Point cloud compression
RS	Remote sensing
LiDAR	Light detection and ranging
RADAR	Radio detection and ranging
SAR	Synthetic aperture RADAR
SONAR	Sound detection and ranging
DL	Deep learning
NeRF	Neural radiance field
SS	Semantic segmentation
SfM	Structure from motion
RGB-D	RGB plus depth

MLS	Mobile laser scanner
PCSS	Point cloud semantic segmentation
UAV	Unmanned aerial vehicle
UAS	Unmanned aircraft system
ALS	Aerial laser scanner
TLS	Terrestrial laser scanner
LAI	Leaf area index
fCover	Vegetation cover fraction
CNN	Convolutional neural network
IS	Instance segmentation
PS	Panoptic segmentation
VR/AR	Virtual reality/augmented reality

References

- van Eck, N.J.; Waltman, L. Software survey: VOSviewer, a computer program for bibliometric mapping. *Scientometrics* 2010, 84, 523–538. https://doi.org/10.1007/s11192-009-0146-3.
- Yang, B.; Haala, N.; Dong, Z. Progress and perspectives of point cloud intelligence. *Geo-Spat. Inf. Sci.* 2023, 26, 189–205. https://doi.org/10.1080/10095020.2023.2175478.
- Wang, M.; Yue, G.; Xiong, J.; Tian, S. Intelligent Point Cloud Processing, Sensing, and Understanding. Sensors 2024, 24, 283. https://doi.org/10.3390/s24010283.
- Nurunnabi, A.A.M.; Chen, M.; Xia, Y.; Teferle, F.N.R. Point Cloud Processing with Machine Learning. *Remote Sens.* 2024, Available online: https://www.mdpi.com/journal/remotesensing/special_issues/5H5UA6M872 (accessed on 22 October 2024).
- Yang, B.; Dong, Z.; Liang, F.; Mi, X. Ubiquitous Point Cloud: Theory, Model, and Applications; Imaging Science, CRC Press: Boca Raton, FL, USA, 2025; pp. 1–250. https://doi.org/10.1201/9781003486060.
- Camuffo, E.; Mari, D.; Milani, S. Recent Advancements in Learning Algorithms for Point Clouds: An Updated Overview. *Sensors* 2022, 22, 1357. https://doi.org/10.3390/s22041357.
- Savelonas, M.A.; Veinidis, C.N.; Bartsokas, T.K. Computer Vision and Pattern Recognition for the Analysis of 2D/3D Remote Sensing Data in Geoscience: A Survey. *Remote Sens.* 2022, 14, 6017. https://doi.org/10.3390/rs14236017.
- Roriz, R.; Silva, H.; Dias, F.; Gomes, T. A Survey on Data Compression Techniques for Automotive LiDAR Point Clouds. *Sensors* 2024, 24, 3185. https://doi.org/10.3390/s24103185.
- Martins, N.A.B.; Cruz, L.A.d.S.; Lopes, F. Impact of LiDAR point cloud compression on 3D object detection evaluated on the KITTI dataset. EURASIP J. Image Video Process. 2024, 2024, 15. https://doi.org/10.1186/s13640-024-00633-4.
- Che, E.; Jung, J.; Olsen, M.J. Object recognition, segmentation, and classification of mobile laser scanning point clouds: A state of the art review. *Sensors* 2019, *19*, 810. https://doi.org/10.3390/s19040810.
- 11. Xie, Y.; Tian, J.; Zhu, X.X. Linking Points with Labels in 3D: A Review of Point Cloud Semantic Segmentation. *IEEE Geosci. Remote Sens. Mag.* 2020, *8*, 38–59. https://doi.org/10.1109/MGRS.2019.2937630.
- 12. Yuan, X.; Shi, J.; Gu, L. A review of deep learning methods for semantic segmentation of remote sensing imagery. *Expert Syst. Appl.* **2021**, *169*, 114417. https://doi.org/10.1016/j.eswa.2020.114417.
- Li, N.; Kahler, O.; Pfeifer, N. A Comparison of Deep Learning Methods for Airborne Lidar Point Clouds Classification. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* 2021, 14, 6467–6486. https://doi.org/10.1109/JSTARS.2021.3091389.
- Diab, A.; Kashef, R.; Shaker, A. Deep Learning for LiDAR Point Cloud Classification in Remote Sensing. Sensors 2022, 22, 7868. https://doi.org/10.3390/s22207868.
- 15. Singh, D.P.; Yadav, M. Deep learning-based semantic segmentation of three-dimensional point cloud: A comprehensive review. *Int. J. Remote Sens.* **2024**, *45*, 532–586. https://doi.org/10.1080/01431161.2023.2297177.
- 16. Griffiths, D.; Boehm, J. A Review on deep learning techniques for 3D sensed data classification. *Remote Sens.* **2019**, *11*, 1499. https://doi.org/10.3390/rs11121499.
- Adam, J.M.; Liu, W.; Zang, Y.; Afzal, M.K.; Bello, S.A.; Muhammad, A.U.; Wang, C.; Li, J. Deep learning-based semantic segmentation of urban-scale 3D meshes in remote sensing: A survey. *Int. J. Appl. Earth Obs. Geoinf.* 2023, 121, 103365. https://doi.org/10.1016/j.jag.2023.103365.
- Cheng, L.; Chen, S.; Liu, X.; Xu, H.; Wu, Y.; Li, M.; Chen, Y. Registration of laser scanning point clouds: A review. Sensors 2018, 18, 1641. https://doi.org/10.3390/s18051641.
- 19. Chen, S.; Deng, C.; Zhang, Y.; Wang, Y.; Zhang, Q.; Zhou, Z. A Survey on the Deep Learning-Based Mismatch Removal: Principles and Methods. *IEEE Access* **2023**, *11*, 106877–106897. https://doi.org/10.1109/ACCESS.2023.3320682.

- Cheng, X.; Liu, X.; Li, J.; Zhou, W. Deep learning-based point cloud registration: A comprehensive investigation. *Int. J. Remote Sens.* 2024, 45, 3412–3442. https://doi.org/10.1080/01431161.2024.2343434.
- 21. Zhang, J.; Lin, X. Advances in fusion of optical imagery and LiDAR point cloud applied to photogrammetry and remote sensing. *Int. J. Image Data Fusion* **2017**, *8*, 1–31. https://doi.org/10.1080/19479832.2016.1160960.
- Aasen, H.; Honkavaara, E.; Lucieer, A.; Zarco-Tejada, P.J. Quantitative remote sensing at ultra-high resolution with UAV spectroscopy: A review of sensor technology, measurement procedures, and data correctionworkflows. *Remote Sens.* 2018, 10, 1091. https://doi.org/10.3390/rs10071091.
- 23. Takhtkeshha, N.; Mandlburger, G.; Remondino, F.; Hyyppä, J. Multispectral Light Detection and Ranging Technology and Applications: A Review. *Sensors* **2024**, *24*, 1669. https://doi.org/10.3390/s24051669.
- Geiger, A.; Lenz, P.; Urtasun, R. Are we ready for autonomous driving? the KITTI vision benchmark suite. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Providence, RI, USA, 2012; IEEE Computer Society: Washington, DC, USA, 2012; pp. 3354–3361. https://doi.org/10.1109/CVPR.2012.6248074.
- Guarda, A.; Perry, S. Verification Model Description for JPEG Pleno Learning-Based Point Cloud Coding v1.0; Technical report, ISO/IEC JTC1/SC29/WG11 input document N100367; International Organization for Standardization (ISO), Geneva, Switzerland; Online; 2022.
- Mammou, K.; Chou, P.A.; Flynn, D.; Krivokuća, M.; Nakagami, O.; Sugio, T. *G-PCC Codec Description v2*; Technical report, ISO/IEC JTC1/SC29/WG11 input document N18189; International Organization for Standardization (ISO), Geneva, Switzerland; Marrakech, MA, USA, 2019.
- 27. Lasserre, S.; Taquet, J. A Point Cloud Codec for Lidar Data with Very Low Complexity and Latency; Technical report, ISO/IEC JTC1/SC29/WG7 input document m56477; International Organization for Standardization (ISO), Geneva, Switzerland; 2021.
- 28. Fischler, M.A.; Bolles, R.C. Random sample consensus: A Paradigm for Model Fitting with Applications to Image Analysis and Automated Cartography. *Commun. ACM* **1981**, *24*, 381–395. https://doi.org/10.1145/358669.358692.
- Weinmann, M.; Weinmann, M.; Hinz, S.; Jutzi, B. Fast and automatic image-based registration of TLS data. *ISPRS J. Photogramm. Remote Sens.* 2011, 66, S62–S70. https://doi.org/10.1016/j.isprsjprs.2011.09.010.
- 30. Martínez-Otzeta, J.M.; Rodríguez-Moreno, I.; Mendialdua, I.; Sierra, B. RANSAC for Robotic Applications: A Survey. *Sensors* 2023, 23, 327. https://doi.org/10.3390/s23010327.
- 31. Wang, R.; Peethambaran, J.; Chen, D. LiDAR Point Clouds to 3-D Urban Models: A Review. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* **2018**, 11, 606–627. https://doi.org/10.1109/JSTARS.2017.2781132.
- 32. Xu, Y.; Stilla, U. Toward Building and Civil Infrastructure Reconstruction from Point Clouds: A Review on Data and Key Techniques. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* **2021**, *14*, 2857–2885. https://doi.org/10.1109/JSTARS.2021.3060568.
- Rashdi, R.; Martínez-Sánchez, J.; Arias, P.; Qiu, Z. Scanning Technologies to Building Information Modelling: A Review. Infrastructures 2022, 7, 49. https://doi.org/10.3390/infrastructures7040049.
- 34. Guan, H.; Li, J.; Cao, S.; Yu, Y. Use of mobile LiDAR in road information inventory: A review. *Int. J. Image Data Fusion* **2016**, 7, 219–242. https://doi.org/10.1080/19479832.2016.1188860.
- Chen, Z.; Deng, L.; Luo, Y.; Li, D.; Marcato Junior, J.; Nunes Gonçalves, W.; Awal Md Nurunnabi, A.; Li, J.; Wang, C.; Li, D. Road extraction in remote sensing data: A survey. *Int. J. Appl. Earth Obs. Geoinf.* 2022, 112, 102833. https://doi.org/10.1016/j.jag.2022.102833.
- Munir, N.; Awrangjeb, M.; Stantic, B. Power Line Extraction and Reconstruction Methods from Laser Scanning Data: A Literature Review. *Remote Sens.* 2023, 15, 973. https://doi.org/10.3390/rs15040973.
- Xiao, W.; Cao, H.; Tang, M.; Zhang, Z.; Chen, N. 3D urban object change detection from aerial and terrestrial point clouds: A review. *Int. J. Appl. Earth Obs. Geoinf.* 2023, 118, 103258. https://doi.org/10.1016/j.jag.2023.103258.
- Kerle, N.; Nex, F.; Gerke, M.; Duarte, D.; Vetrivel, A. UAV-based structural damage mapping: A review. *ISPRS Int. J. Geo-Inf.* 2019, 9, 14. https://doi.org/10.3390/ijgi9010014.
- Sharifisoraki, Z.; Dey, A.; Selzler, R.; Amini, M.; Green, J.; Rajan, S.; Kwamena, F. Monitoring Critical Infrastructure Using 3D LiDAR Point Clouds. *IEEE Access* 2023, 11, 314–336. https://doi.org/10.1109/ACCESS.2022.3232338.
- Rahman, M.K.; Crawford, T.W.; Islam, M.S. Shoreline Change Analysis along Rivers and Deltas: A Systematic Review and Bibliometric Analysis of the Shoreline Study Literature from 2000 to 2021. *Geosciences* 2022, 12, 410. https://doi.org/10.3390/ geosciences12110410.
- 41. Wang, J.; Wang, L.; Feng, S.; Peng, B.; Huang, L.; Fatholahi, S.N.; Tang, L.; Li, J. An Overview of Shoreline Mapping by Using Airborne LiDAR. *Remote Sens.* **2023**, *15*, 253. https://doi.org/10.3390/rs15010253.
- 42. Lissak, C.; Bartsch, A.; De Michele, M.; Gomez, C.; Maquaire, O.; Raucoules, D.; Roulland, T. Remote Sensing for Assessing Landslides and Associated Hazards. *Surv. Geophys.* **2020**, *41*, 1391–1435. https://doi.org/10.1007/s10712-020-09609-1.
- 43. Daghigh, H.; Tannant, D.D.; Daghigh, V.; Lichti, D.D.; Lindenbergh, R. A critical review of discontinuity plane extraction from 3D point cloud data of rock mass surfaces. *Comput. Geosci.* **2022**, *169*, 105241. https://doi.org/10.1016/j.cageo.2022.105241.

- Yan, W.Y.; Shaker, A.; El-Ashmawy, N. Urban land cover classification using airborne LiDAR data: A review. *Remote Sens. Environ.* 2015, 158, 295–310. https://doi.org/10.1016/j.rse.2014.11.001.
- Cotella, V.A. From 3D point clouds to HBIM: Application of Artificial Intelligence in Cultural Heritage. *Autom. Constr.* 2023, 152, 104936. https://doi.org/10.1016/j.autcon.2023.104936.
- Abarca, H.; Deen, R.; Hollins, G.; Zamani, P.; Maki, J.; Tinio, A.; Pariser, O.; Ayoub, F.; Toole, N.; Algermissen, S.; et al. Image and Data Processing for InSight Lander Operations and Science. *Space Sci. Rev.* 2019, 215, 22. https://doi.org/10.1007/s11214-019-0 587-9.
- Amani, M.; Moghimi, A.; Mirmazloumi, S.M.; Ranjgar, B.; Ghorbanian, A.; Ojaghi, S.; Ebrahimy, H.; Naboureh, A.; Nazari, M.E.; Mahdavi, S.; et al. Ocean Remote Sensing Techniques and Applications: A Review (Part I). *Water* 2022, *14*, 3400. https://doi.org/10.3390/w14213400.
- Amani, M.; Mehravar, S.; Asiyabi, R.M.; Moghimi, A.; Ghorbanian, A.; Ahmadi, S.A.; Ebrahimy, H.; Moghaddam, S.H.A.; Naboureh, A.; Ranjgar, B.; et al. Ocean Remote Sensing Techniques and Applications: A Review (Part II). *Water* 2022, 14, 3401. https://doi.org/10.3390/w14213401.
- Chen, W.; Chen, P.; Zhang, H.; He, Y.; Tang, J.; Wu, S. Review of airborne oceanic lidar remote sensing. *Intell. Mar. Technol. Syst.* 2023, 1, 10. https://doi.org/10.1007/s44295-023-00007-y.
- 50. Wang, R.; Wang, J.; Li, Y.; Ma, T.; Zhang, X. Research Advances and Prospects of Underwater Terrain-Aided Navigation. *Remote Sens.* **2024**, *16*, 2560. https://doi.org/10.3390/rs16142560.
- 51. Almeida, L.; Menezes, P.; Dias, J. Telepresence Social Robotics towards Co-Presence: A Review. *Appl. Sci.* 2022, *12*, 5557. https://doi.org/10.3390/app12115557.
- 52. Wang, P.; Wang, Y.; Billinghurst, M.; Yang, H.; Xu, P.; Li, Y. BeHere: A VR/SAR remote collaboration system based on virtual replicas sharing gesture and avatar in a procedural task. *Virtual Real.* **2023**, *27*, 1409–1430. https://doi.org/10.1007/s10055-023-0 0748-5.
- Zhao, L.; Ma, K.K.; Liu, Z.; Yin, Q.; Chen, J. Real-Time Scene-Aware LiDAR Point Cloud Compression Using Semantic Prior Representation. *IEEE Trans. Circuits Syst. Video Technol.* 2022, 32, 5623–5637. https://doi.org/10.1109/TCSVT.2022.3145513.
- Sun, X.; Ma, H.; Sun, Y.; Liu, M. A Novel Point Cloud Compression Algorithm Based on Clustering. *IEEE Robot. Autom. Lett.* 2019, 4, 2132–2139. https://doi.org/10.1109/LRA.2019.2900747.
- 55. Rusu, R.B.; Cousins, S. 3D is here: Point Cloud Library (PCL). In Proceedings of the IEEE International Conference on Robotics and Automation (ICRA), Shanghai, China, 9–13 May 2011.
- 56. Isenburg, M. LASzip: Lossless compression of lidar data. *Photogramm. Eng. Remote Sens.* 2013, 79, 209–217. https://doi.org/10.1 4358/PERS.79.2.209.
- 57. Mongus, D.; Žalik, B. Efficient method for lossless LIDAR data compression. *Int. J. Remote Sens.* **2011**, *32*, 2507–2518. https://doi.org/10.1080/01431161003698385.
- 58. Lipuš, B.; Žalik, B. Lossy LAS file compression using uniform space division. *Electron. Lett.* **2012**, *48*, 1278–1279. https://doi.org/10.1049/el.2012.1080.
- 59. Elseberg, J.; Borrmann, D.; Nüchter, A. One billion points in the cloud—An octree for efficient processing of 3D laser scans. *ISPRS J. Photogramm. Remote Sens.* 2013, 76, 76–88. https://doi.org/10.1016/j.isprsjprs.2012.10.004.
- Hassler, S.C.; Baysal-Gurel, F. Unmanned aircraft system (UAS) technology and applications in agriculture. *Agronomy* 2019, 9, 618. https://doi.org/10.3390/agronomy9100618.
- 61. Rivera, G.; Porras, R.; Florencia, R.; Sánchez-Solís, J.P. LiDAR applications in precision agriculture for cultivating crops: A review of recent advances. *Comput. Electron. Agric.* 2023, 207, 107737. https://doi.org/10.1016/j.compag.2023.107737.
- 62. Lindberg, E.; Holmgren, J. Individual Tree Crown Methods for 3D Data from Remote Sensing. *Curr. For. Rep.* 2017, *3*, 19–31. https://doi.org/10.1007/s40725-017-0051-6.
- Kangas, A.; Astrup, R.; Breidenbach, J.; Fridman, J.; Gobakken, T.; Korhonen, K.T.; Maltamo, M.; Nilsson, M.; Nord-Larsen, T.; Næsset, E.; et al. Remote sensing and forest inventories in Nordic countries–roadmap for the future. *Scand. J. For. Res.* 2018, 33, 397–412. https://doi.org/10.1080/02827581.2017.1416666.
- Kellner, J.R.; Armston, J.; Birrer, M.; Cushman, K.; Duncanson, L.; Eck, C.; Falleger, C.; Imbach, B.; Král, K.; Krůček, M.; et al. New Opportunities for Forest Remote Sensing Through Ultra-High-Density Drone Lidar. *Surv. Geophys.* 2019, 40, 959–977. https://doi.org/10.1007/s10712-019-09529-9.
- Iglhaut, J.; Cabo, C.; Puliti, S.; Piermattei, L.; O'Connor, J.; Rosette, J. Structure from Motion Photogrammetry in Forestry: A Review. Curr. For. Rep. 2019, 5, 155–168. https://doi.org/10.1007/s40725-019-00094-3.
- 66. Kivinen, S.; Koivisto, E.; Keski-Saari, S.; Poikolainen, L.; Tanhuanpää, T.; Kuzmin, A.; Viinikka, A.; Heikkinen, R.K.; Pykälä, J.; Virkkala, R.; et al. A keystone species, European aspen (*Populus tremula* L.), in boreal forests: Ecological role, knowledge needs and mapping using remote sensing. *For. Ecol. Manag.* 2020, *462*, 118008. https://doi.org/10.1016/j.foreco.2020.118008.
- 67. Michałowska, M.; Rapiński, J. A review of tree species classification based on airborne lidar data and applied classifiers. *Remote Sens.* **2021**, *13*, 353. https://doi.org/10.3390/rs13030353.

- 68. Alvites, C.; Marchetti, M.; Lasserre, B.; Santopuoli, G. LiDAR as a Tool for Assessing Timber Assortments: A Systematic Literature Review. *Remote Sens.* **2022**, *14*, 4466. https://doi.org/10.3390/rs14184466.
- Demol, M.; Verbeeck, H.; Gielen, B.; Armston, J.; Burt, A.; Disney, M.; Duncanson, L.; Hackenberg, J.; Kükenbrink, D.; Lau, A.; et al. Estimating forest above-ground biomass with terrestrial laser scanning: Current status and future directions. *Methods Ecol. Evol.* 2022, *13*, 1628–1639. https://doi.org/10.1111/2041-210X.13906.
- Rudge, M.L.M.; Levick, S.R.; Bartolo, R.E.; Erskine, P.D. Developing landscape-scale forest restoration targets that embrace spatial pattern. *Landsc. Ecol.* 2022, 37, 1747–1760. https://doi.org/10.1007/s10980-022-01461-5.
- Yun, T.; Li, J.; Ma, L.; Zhou, J.; Wang, R.; Eichhorn, M.P.; Zhang, H. Status, advancements and prospects of deep learning methods applied in forest studies. *Int. J. Appl. Earth Obs. Geoinf.* 2024, 131, 103938. https://doi.org/10.1016/j.jag.2024.103938.
- Łopatka, A.; Miturski, T.; Pudełko, R.; Kozyra, J.; Koza, P. Review of soil moisture and plant water stress models based on satellite thermal imagery. *Pol. J. Soil Sci.* 2016, 49, 73–90. https://doi.org/10.17951/pjss.2016.49.1.73.
- 73. Tian, L.; Qu, Y.; Qi, J. Estimation of forest lai using discrete airborne lidar: A review. *Remote Sens.* 2021, *13*, 2408. https://doi.org/10.3390/rs13122408.
- 74. Muumbe, T.P.; Baade, J.; Singh, J.; Schmullius, C.; Thau, C. Terrestrial laser scanning for vegetation analyses with a special focus on savannas. *Remote Sens.* **2021**, *13*, 507. https://doi.org/10.3390/rs13030507.
- 75. Li, L.; Mu, X.; Jiang, H.; Chianucci, F.; Hu, R.; Song, W.; Qi, J.; Liu, S.; Zhou, J.; Chen, L.; et al. Review of ground and aerial methods for vegetation cover fraction (fCover) and related quantities estimation: Definitions, advances, challenges, and future perspectives. *ISPRS J. Photogramm. Remote Sens.* 2023, 199, 133–156. https://doi.org/10.1016/j.isprsjprs.2023.03.020.
- Ferro, M.V.; Catania, P. Technologies and Innovative Methods for Precision Viticulture: A Comprehensive Review. *Horticulturae* 2023, *9*, 399. https://doi.org/10.3390/horticulturae9030399.
- 77. Dobbs, A.M.; Ginn, D.; Skovsen, S.K.; Bagavathiannan, M.V.; Mirsky, S.B.; Reberg-Horton, C.S.; Leon, R.G. New directions in weed management and research using 3D imaging. Weed Sci. 2022, 70, 641–647. https://doi.org/10.1017/wsc.2022.56.
- Nex, F.; Stathopoulou, E.; Remondino, F.; Yang, M.; Madhuanand, L.; Yogender, Y.; Alsadik, B.; Weinmann, M.; Jutzi, B.; Qin, R. UseGeo—A UAV-based multi-sensor dataset for geospatial research. *ISPRS Open J. Photogramm. Remote Sens.* 2024, 13. https://doi.org/10.1016/j.ophoto.2024.100070.
- 79. Lingemann, K.; Nüchter, A. Robotic 3D Scan Repository. Available online: http://kos.informatik.uni-osnabrueck.de/3Dscans/ (accessed on 22 October 2024).
- Hackel, T.; Savinov, N.; Ladicky, L.; Wegner, J.; Schindler, K.; Pollefeys, M. SEMANTIC3D.NET: A new large-scale point cloud classification benchmark. In Proceedings of the ISPRS Annals of the Photogrammetry, Remote Sensing and Spatial Information Sciences: Hannover, Germany, 2017; Copernicus GmbH: Göttingen, Germany, 2017; Volume IV-1/W1, pp. 91–98. https://doi.org/10.5194/isprs-annals-IV-1-W1-91-2017.
- Tan, W.; Qin, N.; Ma, L.; Li, Y.; Du, J.; Cai, G.; Yang, K.; Li, J. Toronto-3D: A large-scale mobile LiDAR dataset for semantic segmentation of urban roadways. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition Workshops: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 797–806. https://doi.org/10.1109/CVPRW50498.2020.00109.
- Varney, N.; Asari, V.K.; Graehling, Q. DALES: A large-scale aerial LiDAR data set for semantic segmentation. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition Workshops: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 717–726. https://doi.org/10.1109/CVPRW50498.2020.00101.
- Singer, N.M.; Asari, V.K. DALES Objects: A Large Scale Benchmark Dataset for Instance Segmentation in Aerial Lidar. *IEEE Access* 2021, *9*, 97495–97504. https://doi.org/10.1109/ACCESS.2021.3094127.
- Can, G.; Mantegazza, D.; Abbate, G.; Chappuis, S.; Giusti, A. Semantic segmentation on Swiss3DCities: A benchmark study on aerial photogrammetric 3D pointcloud dataset. *Pattern Recognit. Lett.* 2021, 150, 108–114. https://doi.org/10.1016/j.patrec.2021.0 6.004.
- Hu, Q.; Yang, B.; Khalid, S.; Xiao, W.; Trigoni, N.; Markham, A. Towards Semantic Segmentation of Urban-Scale 3D Point Clouds: A Dataset, Benchmarks and Challenges. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Nashville, TN, USA, 2021; IEEE Computer Society: Washington, DC, USA, 2021; pp. 4975–4985. https://doi.org/10.1109/CVPR46437.2021.00494.
- Selvaraju, P.; Nabail, M.; Loizou, M.; Maslioukova, M.; Averkiou, M.; Andreou, A.; Chaudhuri, S.; Kalogerakis, E. BuildingNet: Learning to Label 3D Buildings. In Proceedings of the IEEE International Conference on Computer Vision: Montreal, BC, Canada, 2021; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2021; pp. 10377–10387. https: //doi.org/10.1109/ICCV48922.2021.01023.
- Chen, M.; Hu, Q.; Yu, Z.; Thomas, H.; Feng, A.; Hou, Y.; McCullough, K.; Ren, F.; Soibelman, L. STPLS3D: A Large-Scale Synthetic and Real Aerial Photogrammetry 3D Point Cloud Dataset. In Proceedings of the BMVC 2022—33rd British Machine Vision Conference Proceedings: London, UK, 2022; British Machine Vision Association, BMVA: Glasgow, UK, 2022.

- Pandey, G.; McBride, J.R.; Eustice, R.M. Ford Campus vision and lidar data set. Int. J. Robot. Res. 2011, 30, 1543–1552. https://doi.org/10.1177/0278364911400640.
- Chang, M.F.; Lambert, J.; Sangkloy, P.; Singh, J.; Bak, S.; Hartnett, A.; Wang, D.; Carr, P.; Lucey, S.; Ramanan, D.; et al. Argoverse: 3D tracking and forecasting with rich maps. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Long Beach, CA, USA, 2019; IEEE Computer Society: Washington, DC, USA, 2019; pp. 8740–8749. https://doi.org/10.1109/CVPR.2019.00895.
- Caesar, H.; Bankiti, V.; Lang, A.H.; Vora, S.; Liong, V.E.; Xu, Q.; Krishnan, A.; Pan, Y.; Baldan, G.; Beijbom, O. Nuscenes: A multimodal dataset for autonomous driving. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 11618–11628. https://doi.org/10.1109/CVPR42600.2020.01164.
- Behley, J.; Garbade, M.; Milioto, A.; Quenzel, J.; Behnke, S.; Gall, J.; Stachniss, C. Towards 3D LiDAR-based semantic scene understanding of 3D point cloud sequences: The SemanticKITTI Dataset. *Int. J. Robot. Res.* 2021, 40, 959–967. https: //doi.org/10.1177/02783649211006735.
- 92. Sun, P.; Kretzschmar, H.; Dotiwalla, X.; Chouard, A.; Patnaik, V.; Tsui, P.; Guo, J.; Zhou, Y.; Chai, Y.; Caine, B.; et al. Scalability in perception for autonomous driving: Waymo open dataset. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 2443–2451. https://doi.org/10.1109/CVPR42600.2020.00252.
- 93. Wilson, B.; Qi, W.; Agarwal, T.; Lambert, J.; Singh, J.; Khandelwal, S.; Pan, B.; Kumar, R.; Hartnett, A.; Pontes, J.K.; et al. Argoverse 2: Next Generation Datasets for Self-driving Perception and Forecasting. In Proceedings of the Neural Information Processing Systems Track on Datasets and Benchmarks (NeurIPS Datasets and Benchmarks 2021): Online, 2021; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2021.
- 94. Mao, J.; Niu, M.; Jiang, C.; Liang, X.; Li, Y.; Ye, C.; Zhang, W.; Li, Z.; Yu, J.; Xu, C. One Million Scenes for Autonomous Driving: ONCE Dataset. In Proceedings of the Neural Information Processing Systems Track on Datasets and Benchmarks (NeurIPS Datasets and Benchmarks 2021): Online, 2021; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2021.
- Xiao, P.; Shao, Z.; Hao, S.; Zhang, Z.; Chai, X.; Jiao, J.; Li, Z.; Wu, J.; Sun, K.; Jiang, K.; et al. PandaSet: Advanced Sensor Suite Dataset for Autonomous Driving. In Proceedings of the IEEE Conference on Intelligent Transportation Systems, Proceedings, ITSC: Indianapolis, IN, USA, 2021; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2021; pp. 3095–3101. https://doi.org/10.1109/ITSC48978.2021.9565009.
- Fong, W.K.; Mohan, R.; Hurtado, J.V.; Zhou, L.; Caesar, H.; Beijbom, O.; Valada, A. Panoptic Nuscenes: A Large-Scale Benchmark for LiDAR Panoptic Segmentation and Tracking. *IEEE Robot. Autom. Lett.* 2022, 7, 3795–3802. https://doi.org/10.1109/LRA.2022 .3148457.
- Yu, H.; Luo, Y.; Shu, M.; Huo, Y.; Yang, Z.; Shi, Y.; Guo, Z.; Li, H.; Hu, X.; Yuan, J.; et al. DAIR-V2X: A Large-Scale Dataset for Vehicle-Infrastructure Cooperative 3D Object Detection. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: New Orleans, LA, USA, 2022; IEEE Computer Society: Washington, DC, USA, 2022; pp. 21329–21338. https://doi.org/10.1109/CVPR52688.2022.02067.
- Yan, F.; Nie, M.; Cai, X.; Han, J.; Xu, H.; Yang, Z.; Ye, C.; Fu, Y.; Mi, M.B.; Zhang, L. ONCE-3DLanes: Building Monocular 3D Lane Detection. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: New Orleans, LA, USA, 2022; IEEE Computer Society: Washington, DC, USA, 2022; pp. 17122–17131. https://doi.org/10.1109/CVPR52688.202 2.01663.
- Diaz-Ruiz, C.A.; Xia, Y.; You, Y.; Nino, J.; Chen, J.; Monica, J.; Chen, X.; Luo, K.; Wang, Y.; Emond, M.; et al. Ithaca365: Dataset and Driving Perception under Repeated and Challenging Weather Conditions. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: New Orleans, LA, USA, 2022; IEEE Computer Society: Washington, DC, USA, 2022; pp. 21351–21360. https://doi.org/10.1109/CVPR52688.2022.02069.
- 100. Li, Y.; Ma, D.; An, Z.; Wang, Z.; Zhong, Y.; Chen, S.; Feng, C. V2X-Sim: Multi-Agent Collaborative Perception Dataset and Benchmark for Autonomous Driving. *IEEE Robot. Autom. Lett.* **2022**, *7*, 10914–10921. https://doi.org/10.1109/LRA.2022.3192802.
- Chen, L.; Sima, C.; Li, Y.; Zheng, Z.; Xu, J.; Geng, X.; Li, H.; He, C.; Shi, J.; Qiao, Y.; et al. PersFormer: 3D Lane Detection via Perspective Transformer and the OpenLane Benchmark. In Proceedings of the Computer Vision—ECCV 2022: Tel Aviv, Israel, 2022; Springer Nature: Cham, Switzerland, 2022; Volume 13698 LNCS, pp. 550–567. https://doi.org/10.1007/978-3-031-19839-7_ 32.
- 102. Wang, H.; Li, T.; Li, Y.; Chen, L.; Sima, C.; Liu, Z.; Wang, B.; Jia, P.; Wang, Y.; Jiang, S.; et al. OpenLane-V2: A Topology Reasoning Benchmark for Unified 3D HD Mapping. In Proceedings of the Advances in Neural Information Processing Systems: New Orleans, LA, USA, 2023; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2023; Volume 36.
- 103. Yu, H.; Yang, W.; Ruan, H.; Yang, Z.; Tang, Y.; Gao, X.; Hao, X.; Shi, Y.; Pan, Y.; Sun, N.; et al. V2X-Seq: A Large-Scale Sequential Dataset for Vehicle-Infrastructure Cooperative Perception and Forecasting. In Proceedings of the 2023 IEEE/CVF Conference on

Computer Vision and Pattern Recognition (CVPR): Vancouver, BC, Canada, 2023; IEEE Computer Society: Washington, DC, USA, 2023; pp. 5486–5495. https://doi.org/10.1109/CVPR52729.2023.00531.

- 104. Xu, R.; Xia, X.; Li, J.; Li, H.; Zhang, S.; Tu, Z.; Meng, Z.; Xiang, H.; Dong, X.; Song, R.; et al. V2V4Real: A Real-World Large-Scale Dataset for Vehicle-to-Vehicle Cooperative Perception. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Vancouver, BC, Canada, 2023; IEEE Computer Society: Washington, DC, USA, 2023; pp. 13712–13722. https://doi.org/10.1109/CVPR52729.2023.01318.
- 105. Tian, X.; Jiang, T.; Yun, L.; Mao, Y.; Yang, H.; Wang, Y.; Wang, Y.; Zhao, H. Occ3D: A Large-Scale 3D Occupancy Prediction Benchmark for Autonomous Driving. In Proceedings of the Advances in Neural Information Processing Systems: New Orleans, LA, USA, 2023; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2023; Volume 36.
- 106. Xiang, H.; Zheng, Z.; Xia, X.; Xu, R.; Gao, L.; Zhou, Z.; Han, X.; Ji, X.; Li, M.; Meng, Z.; et al. V2X-Real: A Largs-Scale Dataset for Vehicle-to-Everything Cooperative Perception. In Proceedings of the Computer Vision—ECCV 2024: Milan, Italy, 2024; Springer Nature: Cham, Switzerland, 2024.
- 107. Yao, S.; Guan, R.; Wu, Z.; Ni, Y.; Huang, Z.; Liu, R.W.; Yue, Y.; Ding, W.; Lim, E.G.; Seo, H.; et al. WaterScenes: A Multi-Task 4D Radar-Camera Fusion Dataset and Benchmarks for Autonomous Driving on Water Surfaces. *IEEE Trans. Intell. Transp. Syst.* 2024, 25, 16584–16598. https://doi.org/10.1109/TITS.2024.3415772.
- 108. Song, S.; Lichtenberg, S.P.; Xiao, J. SUN RGB-D: A RGB-D scene understanding benchmark suite. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Boston, MA, USA, 2015; IEEE Computer Society: Washington, DC, USA, 2015; pp. 567–576. https://doi.org/10.1109/CVPR.2015.7298655.
- 109. Armeni, I.; Sener, O.; Zamir, A.R.; Jiang, H.; Brilakis, I.; Fischer, M.; Savarese, S. 3D semantic parsing of large-scale indoor spaces. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Las Vegas, NV, USA, 2016; IEEE Computer Society: Washington, DC, USA, 2016; pp. 1534–1543. https://doi.org/10.1109/CVPR.2016.170.
- 110. Hua, B.S.; Pham, Q.H.; Nguyen, D.T.; Tran, M.K.; Yu, L.F.; Yeung, S.K. SceneNN: A scene meshes dataset with aNNotations. In Proceedings of the 2016 4th International Conference on 3D Vision, 3DV 2016: Stanford, CA, USA, 2016; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2016; pp. 92–101. https://doi.org/10.1109/3DV.2016.18.
- 111. Dai, A.; Chang, A.X.; Savva, M.; Halber, M.; Funkhouser, T.; Nießner, M. ScanNet: Richly-annotated 3D reconstructions of indoor scenes. In Proceedings of the 30th IEEE Conference on Computer Vision and Pattern Recognition, CVPR 2017: Honolulu, HI, USA, 2017; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2017; pp. 2432–2443. https://doi.org/10.1109/CVPR.2017.261.
- 112. Song, S.; Yu, F.; Zeng, A.; Chang, A.X.; Savva, M.; Funkhouser, T. Semantic scene completion from a single depth image. In Proceedings of the 30th IEEE Conference on Computer Vision and Pattern Recognition, CVPR 2017: Honolulu, HI, USA, 2017; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2017; pp. 190–198. https://doi.org/10.1109/CVPR.20 17.28.
- 113. Zheng, J.; Zhang, J.; Li, J.; Tang, R.; Gao, S.; Zhou, Z. Structured3D: A Large Photo-Realistic Dataset for Structured 3D Modeling. In Proceedings of the Computer Vision—ECCV 2020: Glasgow, UK, 2020; Springer Nature: Cham, Switzerland, 2020; Volume 12354 LNCS, pp. 519–535. https://doi.org/10.1007/978-3-030-58545-7_30.
- 114. Baruch, G.; Chen, Z.; Dehghan, A.; Dimry, T.; Feigin, Y.; Fu, P.; Gebauer, T.; Joffe, B.; Kurz, D.; Schwartz, A.; et al. ARKitScenes—A Diverse Real-World Dataset for 3D Indoor Scene Understanding Using Mobile RGB-D Data. In Proceedings of the Thirtyfifth Conference on Neural Information Processing Systems Datasets and Benchmarks Track (Round 1): Online, 2021; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2021.
- 115. Ramakrishnan, S.K.; Gokaslan, A.; Wijmans, E.; Maksymets, O.; Clegg, A.; Turner, J.M.; Undersander, E.; Galuba, W.; Westbury, A.; Chang, A.X.; et al. Habitat-Matterport 3D Dataset (HM3D): 1000 Large-scale 3D Environments for Embodied AI. In Proceedings of the Thirty-fifth Conference on Neural Information Processing Systems Datasets and Benchmarks Track (Round 2): Online, 2021; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2021.
- 116. Rozenberszki, D.; Litany, O.; Dai, A. Language-Grounded Indoor 3D Semantic Segmentation in the Wild. In Proceedings of the Computer Vision—ECCV 2022: Tel Aviv, Israel, 2022; Springer Nature: Cham, Switzerland, 2022; Volume 13693 LNCS, pp. 125–141. https://doi.org/10.1007/978-3-031-19827-4_8.
- 117. Xu, M.; Chen, P.; Liu, H.; Han, X. TO-Scene: A Large-Scale Dataset for Understanding 3D Tabletop Scenes. In Proceedings of the Computer Vision—ECCV 2022: Tel Aviv, Israel, 2022; Springer Nature: Cham, Switzerland, 2022; Volume 13687 LNCS, pp. 340–356. https://doi.org/10.1007/978-3-031-19812-0_20.
- 118. Yeshwanth, C.; Liu, Y.C.; Niesner, M.; Dai, A. ScanNet++: A High-Fidelity Dataset of 3D Indoor Scenes. In Proceedings of the IEEE International Conference on Computer Vision: Paris, France, 2023; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2023; pp. 12–22. https://doi.org/10.1109/ICCV51070.2023.00008.
- 119. Yadav, K.; Ramrakhya, R.; Ramakrishnan, S.K.; Gervet, T.; Turner, J.; Gokaslan, A.; Maestre, N.; Chang, A.X.; Batra, D.; Savva, M.; et al. Habitat-Matterport 3D Semantics Dataset. In Proceedings of the 2023 IEEE/CVF Conference on Computer Vision and

Pattern Recognition (CVPR): Vancouver, BC, Canada, 2023; IEEE Computer Society: Washington, DC, USA, 2023; pp. 4927–4936. https://doi.org/10.1109/CVPR52729.2023.00477.

- 120. He, S.; Ding, H.; Jiang, X.; Wen, B. SegPoint: Segment Any Point Cloud via Large Language Model. In Proceedings of the Computer Vision—ECCV 2024: Milan, Italy, 2024; Springer Nature: Cham, Switzerland, 2024.
- 121. Wu, Z.; Song, S.; Khosla, A.; Yu, F.; Zhang, L.; Tang, X.; Xiao, J. 3D ShapeNets: A deep representation for volumetric shapes. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Boston, MA, USA, 2015; IEEE Computer Society: Washington, DC, USA, 2015; pp. 1912–1920. https://doi.org/10.1109/CVPR.2015.7298801.
- 122. Chang, A.X.; Funkhouser, T.A.; Guibas, L.J.; Hanrahan, P.; Huang, Q.; Li, Z.; Savarese, S.; Savva, M.; Song, S.; Su, H.; et al. ShapeNet: An Information-Rich 3D Model Repository. *arXiv* 2015, arXiv:1512.03012.
- 123. Yi, L.; Kim, V.G.; Ceylan, D.; Shen, I.C.; Yan, M.; Su, H.; Lu, C.; Huang, Q.; Sheffer, A.; Guibas, L. A scalable active framework for region annotation in 3D shape collections. *ACM Trans. Graph.* **2016**, *35*, 1–12. https://doi.org/10.1145/2980179.2980238.
- 124. Yu, L.; Li, X.; Fu, C.W.; Cohen-Or, D.; Heng, P.A. PU-Net: Point Cloud Upsampling Network. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Salt Lake City, UT, USA, 2018; IEEE Computer Society: Washington, DC, USA, 2018; pp. 2790–2799. https://doi.org/10.1109/CVPR.2018.00295.
- 125. Uy, M.A.; Pham, Q.H.; Hua, B.S.; Nguyen, T.; Yeung, S.K. Revisiting point cloud classification: A new benchmark dataset and classification model on real-world data. In Proceedings of the IEEE International Conference on Computer Vision: Seoul, South Korea, 2019; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2019; pp. 1588–1597. https://doi.org/10.1109/ICCV.2019.00167.
- 126. Mo, K.; Zhu, S.; Chang, A.X.; Yi, L.; Tripathi, S.; Guibas, L.J.; Su, H. Partnet: A large-scale benchmark for fine-grained and hierarchical part-level 3D object understanding. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Long Beach, CA, USA, 2019; IEEE Computer Society: Washington, DC, USA, 2019; pp. 909–918. https://doi.org/10.1109/CVPR.2019.00100.
- 127. Wu, T.; Zhang, J.; Fu, X.; Wang, Y.; Ren, J.; Pan, L.; Wu, W.; Yang, L.; Wang, J.; Qian, C.; et al. OmniObject3D: Large-Vocabulary 3D Object Dataset for Realistic Perception, Reconstruction and Generation. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Vancouver, BC, Canada, 2023; IEEE Computer Society: Washington, DC, USA, 2023; pp. 803–814. https://doi.org/10.1109/CVPR52729.2023.00084.
- 128. Deitke, M.; Liu, R.; Wallingford, M.; Ngo, H.; Michel, O.; Kusupati, A.; Fan, A.; Laforte, C.; Voleti, V.; Gadre, S.Y.; et al. Objaverse-XL: A Universe of 10M+ 3D Objects. In Proceedings of the Advances in Neural Information Processing Systems: New Orleans, LA, USA, 2023; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2023; Volume 36.
- 129. Gené-Mola, J.; Gregorio, E.; Auat Cheein, F.; Guevara, J.; Llorens, J.; Sanz-Cortiella, R.; Escolà, A.; Rosell-Polo, J.R. LFuji-air dataset: Annotated 3D LiDAR point clouds of Fuji apple trees for fruit detection scanned under different forced air flow conditions. *Data Brief* 2020, 29, 105248. https://doi.org/10.1016/j.dib.2020.105248.
- 130. Nelson, H.J.; Papanikolopoulos, N. Learning continuous object representations from point cloud data. In Proceedings of the IEEE International Conference on Intelligent Robots and Systems: Las Vegas, NV, USA, 2020; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2020; pp. 2446–2451. https://doi.org/10.1109/IROS45743.2020.9341765.
- 131. Kölle, M.; Laupheimer, D.; Schmohl, S.; Haala, N.; Rottensteiner, F.; Wegner, J.D.; Ledoux, H. The Hessigheim 3D (H3D) benchmark on semantic segmentation of high-resolution 3D point clouds and textured meshes from UAV LiDAR and Multi-View-Stereo. *ISPRS Open J. Photogramm. Remote Sens.* **2021**, *1*, 100001. https://doi.org/10.1016/j.ophoto.2021.100001.
- 132. Schunck, D.; Magistri, F.; Rosu, R.A.; Cornelißen, A.; Chebrolu, N.; Paulus, S.; Léon, J.; Behnke, S.; Stachniss, C.; Kuhlmann, H.; et al. Pheno4D: A spatio-temporal dataset of maize and tomato plant point clouds for phenotyping and advanced plant analysis. *PLoS ONE* **2021**, *16*, e0256340. https://doi.org/10.1371/journal.pone.0256340.
- 133. Vélez, S.; Ariza-Sentís, M.; Valente, J. VineLiDAR: High-resolution UAV-LiDAR vineyard dataset acquired over two years in northern Spain. *Data Brief* **2023**, *51*, 109686. https://doi.org/10.1016/j.dib.2023.109686.
- 134. Cheng, D.; Cladera, F.; Prabhu, A.; Liu, X.; Zhu, A.; Green, P.C.; Ehsani, R.; Chaudhari, P.; Kumar, V. TreeScope: An Agricultural Robotics Dataset for LiDAR-Based Mapping of Trees in Forests and Orchards. In Proceedings of the IEEE International Conference on Robotics and Automation: Yokohama, Japan, 2024; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2024; pp. 14860–14866. https://doi.org/10.1109/ICRA57147.2024.10611103.
- 135. JPEG Committee. JPEG Pleno Database. Available online: https://jpeg.org/plenodb/ (accessed on 13 September 2020).
- 136. Loop, C.; Cai, Q.; Escolano, S.O.; Chou, P. Microsoft Voxelized Upper Bodies—A Voxelized Point Cloud Dataset; Technical report, ISO/IEC JTC1/SC29 Joint WG11/WG1 (MPEG/JPEG) input document m38673/M72012; International Organization for Standardization (ISO), Geneva, Switzerland; Geneva, Switzerland, 2016. https://jpeg.org/plenodb/pc/microsoft/.
- 137. d'Eon, E.; Harrison, B.; Myers, T.; Chou, P.A. 8i Voxelized Full Bodies—A Voxelized Point Cloud Dataset; Technical report, ISO/IEC JTC1/SC29/WG1 input document M74006 and ISO/IEC JTC1/SC29/WG11 input document m40059; International Organization for Standardization (ISO), Geneva, Switzerland; Geneva, Switzerland, 2017. https://jpeg.org/plenodb/pc/8ilabs/.

- 138. Xu, Y.; Lu, Y.; Wen, Z. Owlii Dynamic Human Mesh Sequence Dataset. Technical Report, ISO/IEC JTC1/SC29/WG11 m41658, Macau, China, 2017. Available online: https://mpeg-pcc.org/index.php/pcc-content-database/owlii-dynamic-human-texturedmesh-sequence-dataset/ (accessed on 22 October 2024).
- Krivokuća, M.; Chou, P.A.; Savill, P. 8i Voxelized Full Bodies—A Voxelized Point Cloud Dataset. Technical Report, ISO/IEC JTC1/SC29/WG1 Input Document m42914, Ljubljana, Slovenia, 2018. Available online: https://mpeg-pcc.org/index.php/pcc-content-database/8i-voxelized-surface-light-field-8ivslf-dataset/ (accessed on 22 October 2024).
- 140. Perry, S.; Cong, H.P.; Da Silva Cruz, L.A.; Prazeres, J.; Pereira, M.; Pinheiro, A.; Dumic, E.; Alexiou, E.; Ebrahimi, T. Quality Evaluation of Static Point Clouds Encoded Using MPEG Codecs. In Proceedings of the International Conference on Image Processing, ICIP: Online, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 3428–3432. https://doi.org/10.1109/ ICIP40778.2020.9191308.
- 141. Graziosi, D.; Nak agami, O.; Kuma, S.; Zaghetto, A.; Suzuki, T.; Tabatabai, A. An overview of ongoing point cloud compression standardization activities: Video-based (V-PCC) and geometry-based (G-PCC). APSIPA Trans. Signal Inf. Process. 2020, 9, e13. https://doi.org/10.1017/ATSIP.2020.12.
- 142. Zerman, E.; Ozcinar, C.; Gao, P.; Smolic, A. Textured Mesh vs Coloured Point Cloud: A Subjective Study for Volumetric Video Compression. In Proceedings of the 2020 12th International Conference on Quality of Multimedia Experience, QoMEX 2020: Athlone, Ireland, 2020; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2020. https://doi.org/10.1109/ QoMEX48832.2020.9123137.
- 143. Google. Draco 3d Data Compression, 2024. Available online: https://github.com/google/draco (accessed on 6 August 2024).
- 144. Yang, Q.; Chen, H.; Ma, Z.; Xu, Y.; Tang, R.; Sun, J. Predicting the Perceptual Quality of Point Cloud: A 3D-to-2D Projection-Based Exploration. *IEEE Trans. Multimed.* **2021**, *23*, 3877–3891. https://doi.org/10.1109/TMM.2020.3033117.
- 145. Gautier, G.; Mercat, A.; Freneau, L.; Pitkanen, M.; Vanne, J. UVG-VPC: Voxelized Point Cloud Dataset for Visual Volumetric Video-based Coding. In Proceedings of the 2023 15th International Conference on Quality of Multimedia Experience, QoMEX 2023: Ghent, Belgium, 2023; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2023; pp. 244–247. https://doi.org/10.1109/QoMEX58391.2023.10178589.
- 146. Ak, A.; Zerman, E.; Quach, M.; Chetouani, A.; Smolic, A.; Valenzise, G.; Le Callet, P. BASICS: Broad Quality Assessment of Static Point Clouds in a Compression Scenario. *IEEE Trans. Multimed.* **2024**, *26*, 6730–6742. https://doi.org/10.1109/TMM.2024.3355642.
- 147. Dumic, E.; da Silva Cruz, L.A. Subjective Quality Assessment of V-PCC-Compressed Dynamic Point Clouds Degraded by Packet Losses. *Sensors* 2023, 23, 5623. https://doi.org/10.3390/s23125623.
- 148. Liu, Q.; Su, H.; Duanmu, Z.; Liu, W.; Wang, Z. Perceptual Quality Assessment of Colored 3D Point Clouds. *IEEE Trans. Vis. Comput. Graph.* 2023, 29, 3642–3655. https://doi.org/10.1109/TVCG.2022.3167151.
- Cruz, L. JPEG Pleno Point Cloud Coding Common Training and Test Conditions v2.2; Technical report, ISO/IEC JTC1/SC29/WG11 input document N100909; International Organization for Standardization (ISO), Geneva, Switzerland; Sapporo, Japan, 2024.
- 150. Dumic, E.; Cruz, L.A.d.S. Point cloud coding solutions, subjective assessment and objective measures: A case study. *Symmetry* **2020**, *12*, 1955. https://doi.org/10.3390/sym12121955.
- 151. Preda, M. Test Model for Geometry-Based Solid Point Cloud—GeS TM 3.0; Technical report, ISO/IEC JTC 1/SC 29/WG 7, w23324; International Organization for Standardization (ISO), Geneva, Switzerland; Hannover, Germany, 2023.
- 152. Mekuria, R.; Blom, K.; Cesar, P. Design, Implementation, and Evaluation of a Point Cloud Codec for Tele-Immersive Video. *IEEE Trans. Circuits Syst. Video Technol.* 2017, 27, 828–842. https://doi.org/10.1109/TCSVT.2016.2543039.
- 153. Point Cloud Compression Requirements and Technology Discussion Group. AVS Point Cloud Compression Requirements and Technical Analysis Report v0.1; Technical report, Output document N2643, AVS, 68th AVS meeting; Audio Video Coding Standard Workgroup of China: Beijing, China; Qingdao, China, 2019..
- 154. Houshiar, H.; Nuchter, A. 3D point cloud compression using conventional image compression for efficient data transmission. In Proceedings of the 2015 25th International Conference on Information, Communication and Automation Technologies, ICAT 2015: Sarajevo, Bosnia and Herzegovina, 2015; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2015. https://doi.org/10.1109/ICAT.2015.7340499.
- 155. Dumic, E.; Bjelopera, A.; Nüchter, A. Dynamic point cloud compression based on projections, surface reconstruction and videompression. *Sensors* **2022**, 22, 197. https://doi.org/10.3390/s22010197.
- 156. Zakharchenko, V. V-PCC Codec Description; Technical report, ISO/IEC JTC1/SC29/WG11 input document N18190; International Organization for Standardization (ISO), Geneva, Switzerland; Marrakech, MA, USA, 2019.
- 157. Milani, S. A Syndrome-Based Autoencoder for Point Cloud Geometry Compression. In Proceedings of the International Conference on Image Processing, ICIP: Online, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 2686–2690. https://doi.org/10.1109/ICIP40778.2020.9190647.

- Milani, S. Adae: Adversarial distributed source autoencoder for point cloud compression. In Proceedings of the International Conference on Image Processing, ICIP: Anchorage, AK, USA, 2021; IEEE Computer Society: Washington, DC, USA, 2021; pp. 3078–3082. https://doi.org/10.1109/ICIP42928.2021.9506750.
- 159. Wang, J.; Zhu, H.; Liu, H.; Ma, Z. Lossy Point Cloud Geometry Compression via End-to-End Learning. *IEEE Trans. Circuits Syst. Video Technol.* 2021, *31*, 4909–4923. https://doi.org/10.1109/TCSVT.2021.3051377.
- Wang, J.; Ding, D.; Li, Z.; Ma, Z. Multiscale Point Cloud Geometry Compression. In Proceedings of the Data Compression Conference Proceedings: Snowbird, UT, USA, 2021; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2021; pp. 73–82. https://doi.org/10.1109/DCC50243.2021.00015.
- Wang, J.; Ding, D.; Li, Z.; Feng, X.; Cao, C.; Ma, Z. Sparse Tensor-Based Multiscale Representation for Point Cloud Geometry Compression. *IEEE Trans. Pattern Anal. Mach. Intell.* 2023, 45, 9055–9071. https://doi.org/10.1109/TPAMI.2022.3225816.
- 162. Liu, G.; Wang, J.; Ding, D.; Ma, Z. PCGFormer: Lossy Point Cloud Geometry Compression via Local Self-Attention. In Proceedings of the 2022 IEEE International Conference on Visual Communications and Image Processing, VCIP 2022: Suzhou, China, 2022; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2022. https://doi.org/10.1109/VCIP564 04.2022.10008892.
- 163. Quach, M.; Valenzise, G.; Dufaux, F. Learning Convolutional Transforms for Lossy Point Cloud Geometry Compression. In Proceedings of the International Conference on Image Processing, ICIP: Taipei, Taiwan, 2019; IEEE Computer Society: Washington, DC, USA, 2019; pp. 4320–4324. https://doi.org/10.1109/ICIP.2019.8803413.
- 164. Quach, M.; Valenzise, G.; Dufaux, F. Improved Deep Point Cloud Geometry Compression. In Proceedings of the IEEE 22nd International Workshop on Multimedia Signal Processing, MMSP 2020: Tampere, Finland, 2020; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2020. https://doi.org/10.1109/MMSP48831.2020.9287077.
- 165. Huang, L.; Wang, S.; Wong, K.; Liu, J.; Urtasun, R. OctSqueeze: Octree-structured entropy model for lidar compression. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 1310–1320. https://doi.org/10.1109/CVPR42600.2020.00139.
- 166. Nguyen, D.T.; Quach, M.; Valenzise, G.; Duhamel, P. Learning-based lossless compression of 3D point cloud geometry. In Proceedings of the ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing: Toronto, Canada, 2021; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2021; pp. 4220–4224. https://doi.org/10.1109/ ICASSP39728.2021.9414763.
- 167. Nguyen, D.T.; Quach, M.; Valenzise, G.; Duhamel, P. Multiscale deep context modeling for lossless point cloud geometry compression. In Proceedings of the 2021 IEEE International Conference on Multimedia and Expo Workshops, ICMEW 2021: Shenzhen, China, 2021; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2021. https://doi.org/10.1109/ ICMEW53276.2021.9455990.
- Chen, Z.; Qian, Z.; Wang, S.; Chen, Q. Point Cloud Compression with Sibling Context and Surface Priors. In Proceedings of the Computer Vision—ECCV 2022: Tel Aviv, Israel, 2022; Springer Nature: Cham Switzerland, 2022; Volume 13698 LNCS, pp. 744–759. https://doi.org/10.1007/978-3-031-19839-7_43.
- Cui, M.; Long, J.; Feng, M.; Li, B.; Huang, K. OctFormer: Efficient Octree-Based Transformer for Point Cloud Compression with Local Enhancement. In Proceedings of the 37th AAAI Conference on Artificial Intelligence, AAAI 2023: Washington, DC, USA, 2023; AAAI Press: Washington, DC, USA, 2023; Volume 37, pp. 470–478.
- 170. Zhang, Z.; Zhu, Z.; Bai, Y.; Wang, M.; Yu, Z. Octree-Retention Fusion: A High-Performance Context Model for Point Cloud Geometry Compression. In Proceedings of the 2024 International Conference on Multimedia Retrieval, ICMR 2024: Phuket, Thailand, 2024; Association for Computing Machinery, Inc.: New York, NY, USA, 2024; pp. 1150–1154. https://doi.org/10.1145/ 3652583.3657620.
- 171. Huang, B.; Lazzarotto, D.; Ebrahimi, T. Temporal Conditional Coding for Dynamic Point Cloud Geometry Compression. In Proceedings of the ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing: Seoul, South Korea, 2024; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2024. https://doi.org/10.1109/ICASSP48485.2024.10 446813.
- 172. Fan, T.; Gao, L.; Xu, Y.; Li, Z.; Wang, D. D-DPCC: Deep Dynamic Point Cloud Compression via 3D Motion Prediction. In Proceedings of the IJCAI International Joint Conference on Artificial Intelligence: Vienna, Austria, 2022; pp. 898–904.
- 173. Akhtar, A.; Li, Z.; Van Der Auwera, G. Inter-Frame Compression for Dynamic Point Cloud Geometry Coding. *IEEE Trans. Image Process.* **2024**, *33*, 584–594. https://doi.org/10.1109/TIP.2023.3343096.
- 174. Wang, J.; Ding, D.; Chen, H.; Ma, Z. Dynamic Point Cloud Geometry Compression Using Multiscale Inter Conditional Coding. *arXiv* 2023, arXiv:2301.12165.
- 175. Biswas, S.; Liu, J.; Wong, K.; Wang, S.; Urtasun, R. MuSCLE: Multi sweep compression of LiDAR using deep entropy models. In Proceedings of the Advances in Neural Information Processing Systems: Vancouver, BC, Canada, 2020; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2020.

- 176. Que, Z.; Lu, G.; Xu, D. VoxelContext-Net: An Octree based Framework for Point Cloud Compression. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Nashville, TN, USA, 2021; IEEE Computer Society: Washington, DC, USA, 2021; pp. 6038–6047. https://doi.org/10.1109/CVPR46437.2021.00598.
- 177. Fu, C.; Li, G.; Song, R.; Gao, W.; Liu, S. OctAttention: Octree-Based Large-Scale Contexts Model for Point Cloud Compression. In Proceedings of the 36th AAAI Conference on Artificial Intelligence, AAAI 2022: Online, 2022; Association for the Advancement of Artificial Intelligence: Washington, DC, USA, 2022; Volume 36, pp. 625–633.
- 178. Song, R.; Fu, C.; Liu, S.; Li, G. Large-Scale Spatio-Temporal Attention Based Entropy Model for Point Cloud Compression. In Proceedings of the IEEE International Conference on Multimedia and Expo: Brisbane, Australia, 2023; IEEE Computer Society: Washington, DC, USA, 2023; pp. 2003–2008. https://doi.org/10.1109/ICME55011.2023.00343.
- 179. Song, R.; Fu, C.; Liu, S.; Li, G. Efficient Hierarchical Entropy Model for Learned Point Cloud Compression. In Proceedings of the 2023 IEEE/CVF Conference on Computer Vision and Pattern Recognition (CVPR): Vancouver, BC, Canada, 2023; pp. 14368–14377. https://doi.org/10.1109/CVPR52729.2023.01381.
- 180. Tu, C.; Takeuchi, E.; Carballo, A.; Takeda, K. Point cloud compression for 3d lidar sensor using recurrent neural network with residual blocks. In Proceedings of the IEEE International Conference on Robotics and Automation: Montreal, QC, Canada, 2019; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2019; pp. 3274–3280. https://doi.org/10.1109/ICRA. 2019.8794264.
- 181. Achlioptas, P.; Diamanti, O.; Mitliagkas, I.; Guibas, L. Learning representations and generative models for 3d point clouds. In Proceedings of the 35th International Conference on Machine Learning, ICML 2018: Stockholm, Sweden, 2018; International Machine Learning Society (IMLS): Stroudsburg, PA, USA, 2018; Volume 1, pp. 67–85.
- Mari, D.; Camuffo, E.; Milani, S. CACTUS: Content-Aware Compression and Transmission Using Semantics for Automotive LiDAR Data. Sensors 2023, 23, 5611. https://doi.org/10.3390/s23125611.
- 183. Huang, T.; Zhang, J.; Chen, J.; Ding, Z.; Tai, Y.; Zhang, Z.; Wang, C.; Liu, Y. 3QNet: 3D Point Cloud Geometry Quantization Compression Network. ACM Trans. Graph. 2022, 41, 1–13. https://doi.org/10.1145/3550454.3555481.
- 184. You, K.; Gao, P.; Li, Q. IPDAE: Improved Patch-Based Deep Autoencoder for Lossy Point Cloud Geometry Compression. In Proceedings of the 1st International Workshop on Advances in Point Cloud Compression, Processing and Analysis, APCCPA 2022: Lisbon, Portugal, 2022; Association for Computing Machinery, Inc.: New York, NY, USA, 2022; pp. 1–10. https: //doi.org/10.1145/3552457.3555728.
- 185. You, K.; Liu, K.; Yu, L.; Gao, P.; Ding, D. Pointsoup: High-Performance and Extremely Low-Decoding-Latency Learned Geometry Codec for Large-Scale Point Cloud Scenes. In Proceedings of the Thirty-Third International Joint Conference on Artificial Intelligence, IJCAI-24: Jeju, South Korea, 2024; Larson, K., Ed.; International Joint Conferences on Artificial Intelligence Organization: San Francisco, CA, USA, 2024; pp. 5380–5388. https://doi.org/10.24963/ijcai.2024/595.
- Nguyen, D.T.; Kaup, A. Lossless Point Cloud Geometry and Attribute Compression Using a Learned Conditional Probability Model. *IEEE Trans. Circuits Syst. Video Technol.* 2023, 33, 4337–4348. https://doi.org/10.1109/TCSVT.2023.3239321.
- Wang, J.; Xue, R.; Li, J.; Ding, D.; Lin, Y.; Ma, Z. A Versatile Point Cloud Compressor Using Universal Multiscale Conditional Coding—Part I: Geometry. *IEEE Trans. Pattern Anal. Mach. Intell.* 2024, 47, 269–287. https://doi.org/10.1109/TPAMI.2024.34629 38.
- Wang, J.; Xue, R.; Li, J.; Ding, D.; Lin, Y.; Ma, Z. A Versatile Point Cloud Compressor Using Universal Multiscale Conditional Coding—Part II: Attribute. *IEEE Trans. Pattern Anal. Mach. Intell.* 2024, 47, 252–268. https://doi.org/10.1109/TPAMI.2024.34629 45.
- Sheng, X.; Li, L.; Liu, D.; Xiong, Z.; Li, Z.; Wu, F. Deep-PCAC: An End-to-End Deep Lossy Compression Framework for Point Cloud Attributes. *IEEE Trans. Multimed.* 2022, 24, 2617–2632. https://doi.org/10.1109/TMM.2021.3086711.
- Nguyen, D.T.; Nambiar, K.G.; Kaup, A. Deep Probabilistic Model for Lossless Scalable Point Cloud Attribute Compression. In Proceedings of the ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing: Rhodes Island, Greece, 2023; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2023. https://doi.org/10.1109/ICASSP49357.202 3.10095385.
- Zhang, J.; Wang, J.; Ding, D.; Ma, Z. Scalable Point Cloud Attribute Compression. *IEEE Trans. Multimed.* 2025, 27, 889–899. https://doi.org/10.1109/TMM.2023.3331584.
- 192. Lin, T.P.; Yim, M.; Chiang, J.C.; Peng, W.H.; Lie, W.N. Sparse Tensor-based point cloud attribute compression using Augmented Normalizing Flows. In Proceedings of the 2023 Asia Pacific Signal and Information Processing Association Annual Summit and Conference, APSIPA ASC 2023: Taipei, Taiwan, 2023; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2023; pp. 1739–1744. https://doi.org/10.1109/APSIPAASC58517.2023.10317255.
- 193. You, K.; Gao, P.; Ma, Z. Efficient and Generic Point Model for Lossless Point Cloud Attribute Compression. *arXiv* 2024, arXiv:2404.06936.

- 194. Nguyen, D.T.; Zieger, D.; Stamminger, M.; Kaup, A. End-to-end learned Lossy Dynamic Point Cloud Attribute Compression. In Proceedings of the International Conference on Image Processing, ICIP: Abu Dhabi, United Arab Emirates, 2024; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2024.
- 195. Hu, Y.; Wang, Y. Learning Neural Volumetric Field for Point Cloud Geometry Compression. In Proceedings of the 2022 Picture Coding Symposium, PCS 2022: San Jose, CA, USA, 2022; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2022; pp. 127–131. https://doi.org/10.1109/PCS56426.2022.10018029.
- 196. Isik, B.; Chou, P.A.; Hwang, S.J.; Johnston, N.; Toderici, G. LVAC: Learned Volumetric Attribute Compression for Point Clouds using Coordinate Based Networks. *Front. Signal Process.* **2022**, *2*, 1008812. https://doi.org/10.3389/frsip.2022.1008812.
- 197. Ruan, H.; Shao, Y.; Yang, Q.; Zhao, L.; Niyato, D. Point Cloud Compression with Implicit Neural Representations: A Unified Framework. In Proceedings of the 2024 IEEE/CIC International Conference on Communications in China, ICCC 2024: Hangzhou, China, 2024; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2024.
- 198. Shi, Y.; Zhao, R.; Gasparini, S.; Morin, G.; Ooi, W.T. Volumetric Video Compression Through Neural-based Representation. In Proceedings of the 2024 16th International Workshop on Immersive Mixed and Virtual Environment Systems, MMVE 2024: Bari, Italy, 2024; Association for Computing Machinery, Inc.: New York, NY, USA, 2024; pp. 85–91. https://doi.org/10.1145/3652212. 3652220.
- 199. Umair, S.; Kathariya, B.; Li, Z.; Akhtar, A.; Auwera, G.V.d. ResNeRF-PCAC: Super Resolving Residual Learning NeRF for High Efficiency Point Cloud Attributes Coding. In Proceedings of the 2024 IEEE International Conference on Image Processing (ICIP): Abu Dhabi, United Arab Emirates, 2024; pp. 3540–3546. https://doi.org/10.1109/ICIP51287.2024.10648090.
- 200. Extensis. MrSID Compression. 2024. Available online: https://www.extensis.com/mrsid-file-format (accessed on 3 December 2024).
- 201. Li, G.; Gao, W.; Gao, W. Point Cloud Compression: Technologies and Standardization; Springer Nature: Singapore, 2024; pp. 1–253. https://doi.org/10.1007/978-981-97-1957-0.
- 202. Zhang, W.; Yang, F.; Xu, Y.; Preda, M. Standardization Status of MPEG Geometry-Based Point Cloud Compression (G-PCC) Edition 2. In Proceedings of the 2024 Picture Coding Symposium, PCS 2024: Taichung, Taiwan, 2024; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2024. https://doi.org/10.1109/PCS60826.2024.10566443.
- 203. Zhang, J.; Chen, T.; Ding, D.; Ma, Z. G-PCC++: Enhanced Geometry-based Point Cloud Compression. In Proceedings of the 31st ACM International Conference on Multimedia, MM 2023: Ottawa, ON, Canada, 2023; Association for Computing Machinery, Inc.: New York, NY, USA, 2023; pp. 1352–1363. https://doi.org/10.1145/3581783.3613827.
- Wang, Z.; Wan, S.; Wei, L. Optimized octree codec for geometry-based point cloud compression. *Signal Image Video Process.* 2024, 18, 761–772. https://doi.org/10.1007/s11760-023-02803-9.
- Garcia, D.C.; Fonseca, T.A.; Ferreira, R.U.; De Queiroz, R.L. Geometry Coding for Dynamic Voxelized Point Clouds Using Octrees and Multiple Contexts. *IEEE Trans. Image Process.* 2020, 29, 313–322. https://doi.org/10.1109/TIP.2019.2931466.
- 206. Wang, Y.; Gao, W.; Mu, X.; Yuan, H. Rate Control Optimization for Joint Geometry and Attribute Coding of LiDAR Point Clouds. In Proceedings of the 2023 IEEE International Conference on Visual Communications and Image Processing, VCIP 2023: Jeju, South Korea, 2023; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2023. https://doi.org/10.1109/ VCIP59821.2023.10402779.
- Luo, H.; Chi, Y.; Lu, S.; Ding, Y.; Han, C. Improved Patch Packing and Refining Segmentation for the V-PCC Standard. *Appl. Sci.* 2024, 14, 1405. https://doi.org/10.3390/app14041405.
- Xiong, J.; Gao, H.; Wang, M.; Li, H.; Ngan, K.N.; Lin, W. Efficient Geometry Surface Coding in V-PCC. *IEEE Trans. Multimed.* 2023, 25, 3329–3342. https://doi.org/10.1109/TMM.2022.3158809.
- Que, S.; Li, Y. Lightweight fully connected network-based fast CU size decision for video-based point cloud compression. *Comput. Graph.* 2023, 117, 20–30. https://doi.org/10.1016/j.cag.2023.10.007.
- Tohidi, F.; Paul, M.; Ulhaq, A.; Chakraborty, S. Improved Video-Based Point Cloud Compression via Segmentation. Sensors 2024, 24, 4285. https://doi.org/10.3390/s24134285.
- 211. Luo, A.; Song, L.; Nonaka, K.; Unno, K.; Sun, H.; Goto, M.; Katto, J. SCP: Spherical-Coordinate-Based Learned Point Cloud Compression. In Proceedings of the AAAI Conference on Artificial Intelligence: Vancouver, BC, Canada, 2024; Association for the Advancement of Artificial Intelligence: Washington, DC, USA, 2024; Volume 38, pp. 3954–3962. https://doi.org/10.1609/ aaai.v38i4.28188.
- 212. Wang, J.; Ma, Z. Sparse Tensor-based Point Cloud Attribute Compression. In Proceedings of the 5th International Conference on Multimedia Information Processing and Retrieval, MIPR 2022: Online, 2022; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2022; pp. 59–64. https://doi.org/10.1109/MIPR54900.2022.00018.
- Souto, A.L.; Figueiredo, V.F.; Chou, P.A.; De Queiroz, R.L. Set partitioning in hierarchical trees for point cloud attribute compression. *IEEE Signal Process. Lett.* 2021, 28, 1903–1907. https://doi.org/10.1109/LSP.2021.3112335.
- Figueiredo, V.F.; De Queiroz, R.L.; Chou, P.A.; Lopes, L.S. Embedded Coding of Point Cloud Attributes. *IEEE Signal Process. Lett.* 2024, 31, 890–893. https://doi.org/10.1109/LSP.2024.3378676.

- 215. Guarda, A.; Perry, S. Verification Model Description for JPEG Pleno Learning-Based Point Cloud Coding v4.0; Technical report, ISO/IEC JTC1/SC29/WG11 input document N100709; International Organization for Standardization (ISO), Geneva, Switzerland; San Francisco, CA, USA, 2024.
- Perry, S.; Cruz, L. JPEG White Paper: JPEG Pleno Learning-based Point Cloud Coding v1.0; Technical report, ISO/IEC JTC1/SC29/WG11 input document N100984; International Organization for Standardization (ISO), Geneva, Switzerland; Berlin, Germany, 2024.
- 217. Hu, Q.; Yang, B.; Xie, L.; Rosa, S.; Guo, Y.; Wang, Z.; Trigoni, N.; Markham, A. Randla-Net: Efficient semantic segmentation of large-scale point clouds. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Seattle, WA, USA, 2020; IEEE Computer Society: Washington, DC, USA, 2020; pp. 11105–11114. https://doi.org/10.1109/CVPR4 2600.2020.01112.
- 218. You, K.; Gao, P. Patch-Based Deep Autoencoder for Point Cloud Geometry Compression. In Proceedings of the 3rd ACM International Conference on Multimedia in Asia, MMAsia 21: Gold Coast, Australia, 2021; Association for Computing Machinery: New York, NY, USA, 2021. https://doi.org/10.1145/3469877.3490611.
- 219. Qi, C.R.; Su, H.; Mo, K.; Guibas, L.J. PointNet: Deep learning on point sets for 3D classification and segmentation. In Proceedings of the 30th IEEE Conference on Computer Vision and Pattern Recognition, CVPR 2017: Honolulu, HI, USA, 2017; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2017; pp. 77–85. https://doi.org/10.1109/CVPR.2017.16.
- 220. Mildenhall, B.; Srinivasan, P.P.; Tancik, M.; Barron, J.T.; Ramamoorthi, R.; Ng, R. NeRF: Representing Scenes as Neural Radiance Fields for View Synthesis. In Proceedings of the Computer Vision—ECCV 2020: Glasgow, UK, 2020; Springer Science and Business Media Deutschland GmbH: Berlin, Germany, 2020; Volume 12346 LNCS, pp. 405–421. https://doi.org/10.1007/978-3-030-58452-8_24.
- 221. Shin, S.; Park, J. Binary Radiance Fields. In Proceedings of the Advances in Neural Information Processing Systems: New Orleans, LA, USA, 2023; Neural Information Processing Systems Foundation: La Jolla, CA, USA, 2023; Volume 36, pp. 55919–55931.
- 222. Chen, Y.; Wu, Q.; Harandi, M.; Cai, J. How Far can we Compress Instant-NGP-Based NeRF? In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition: Seattle, WA, USA, 2024; IEEE Computer Society: Washington, DC, USA, 2024; pp. 20321–20330. https://doi.org/10.1109/CVPR52733.2024.01921.
- Knapitsch, A.; Park, J.; Zhou, Q.Y.; Koltun, V. Tanks and temples: Benchmarking large-scale scene reconstruction. ACM Trans. Graph. 2017, 36, 1–13. https://doi.org/10.1145/3072959.3073599.
- 224. Rapidlasso. LASzip—Free and Lossless LiDAR Compression. 2024. Available online: https://laszip.org/ (accessed on 3 December 2024).
- 225. Laboratory for Geometric Modeling and Multimedia Algorithms (GeMMA Lab). LASCompression a Lossless Compression Algorithm for LiDAR Datasets. 2011. Available online: https://gemma.feri.um.si/lascompression/ (accessed on 3 December 2024).
- 226. Otepka, J.; Mandlburger, G.; Schütz, M.; Pfeifer, N.; Wimmer, M. Efficient loading and visualization of massive feature-rich point clouds without hierarchical acceleration structures. In Proceedings of the International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences—ISPRS Archives: Nice, France, 2020; International Society for Photogrammetry and Remote Sensing: Hanover, Germany, 2020; Volume 43, pp. 293–300. https://doi.org/10.5194/isprs-archives-XLIII-B2-2020-293-2020.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.